Technische Universität München TUM School of Social Sciences and Technology



Digital media to enhance learning in informal learning contexts

With a focus on knowledge acquisition, motivational and cognitive aspects especially in the STEM field

Miriam Sarah Lechner

Vollständiger Abdruck der von der TUM School of Social Sciences and Technology

der Technischen Universität München zur Erlangung einer

Doktorin der Philosophie (Dr. phil.)

genehmigten Dissertation.

Vorsitz:

Prof. Dr. Anna Keune

Prüfende der Dissertation: 1. Prof. Dr. Doris Lewalter

2. Prof. Dr. Jenna Koenen

Die Dissertation wurde am 08.04.2024 bei der Technischen Universität München eingereicht und

durch die TUM School of Social Sciences and Technology am 02.07.2024 angenommen.

Dedicated to my beloved daughter Marie "Any day spent with you is my favorite day. So, today is my new favorite day." Winnie the Pooh

Acknowledgments

Writing this doctoral thesis would not have been possible without the support of numerous people. I would like to take this opportunity to thank everyone who has accompanied me on this journey over the last few years and made my time at TUM a very special period of my life.

First of all, I would like to thank my supervisor, Prof. Dr. Doris Lewalter. Dear Doris, first thank you for giving me the opportunity to work on my doctoral thesis at your chair and thank you for always being at my side with help and advice during all the small and large challenges of the last few years, especially in the tricky initial phase of finding a research topic. A huge thank you goes to my mentor Dr. Stephanie Moser. Dearest Stephie, I couldn't have wished for a better mentor than you! Thank you for all the support in all situations during my PhD, for all the things I was able to learn from you and for sweetening the many hours we spent together as desk neighbors. I loved working with you! Another big thank you goes to Prof. Dr. Jürgen Geist and Dr. Joachim Pander, who made my study possible in collaboration with the Chair of Aquatic Systems Biology. Thank you for all the professional expertise and support during the study and in writing the article and for the great help in recruiting participants.

I also want to thank Prof. Dr. Jenna Koenen for sharing her expertise on worked example with me and for taking on the role of second examiner. I would like to thank Prof. Dr. Anna Keune for taking on the role of committee chair. Another thanks goes to Eva Dörfler for her help with the implementation of the Moodle learning environment. Thanks to the entire Chair of Formal and Informal Learning, Dr. Jessica Bodensteiner, Dr. Siëlle Gramser, Dr. Sarah Kellberg, Dr. Katrin Neubauer, Felix Rausch, Jana-Kristin von Wachter and Amina Zerouali for the helpful discussions in the Oberseminar and the many good conversations in the office and via Zoom. And many thanks to the best research assistants Iris Buchmayer, Nalin Hülß, Benedict Ohmann, Marvin Fendt, Adani Abutto, Clara Keil, Stefan Lehner and Johanna Schwarz for their active help in carrying out the study in the field no matter if it was cold and rainy or in the most beautiful sunshine. And another big thanks to my two cousins Colin and Robin Degner for their spontaneous support with my study.

Finally, I would like to thank my family, especially my parents Yvonne and Uwe Degner, as well as my friends for their emotional support and the many great times we have had together over the last few years, which have created a wonderful balance to writing my thesis.

My biggest thanks go to my husband Michael, thank you for being my biggest supporter and motivator and best discussion partner since the beginning of my doctoral thesis, especially when I had to say "mi gira la testa". And thank you to my absolute favorite little person in the world, Marie, for being the most wonderful and sweetest distraction during the challenging final phase of my doctoral thesis.

Abstract

Digital media have a high potential for supporting learning processes in both formal and institutional informal learning places. The present dissertation examines learning with digital media once in an institutional informal context as a systematic literature review and once as an applied example in a formal setting, but in an informal learning context.

In order to get a first overview of the research on the use of digital media in institutional informal learning places as well as connecting points for further research, a systematic literature review (Article A) was conducted on the basis of 26 studies in the first step. Qualitative content analysis was used to examine the identified studies on the general characteristics of digital media and their functions in institutional informal learning places, as well as on the measured outcomes regarding informal learning with digital media in institutional informal learning places. The results show that portable and stationary digital media are used somewhat equally frequently and often include augmented reality. However, the range of possible functions is not yet fully exploited in the conceptualization. It has been shown that digital media are able to enhance motivational and cognitive learning processes, especially in knowledge acquisition and interest as well as in collaboration and social interaction.

Connecting to the findings from the systematic literature review, the second step was an empirical study examining an exemplary use of digital media in a formal setting but in an informal learning context (Article B) with a special focus on how to support learning. In the study, 62 university students used different types of double-content worked examples (WE) integrated into a digital learning environment to learn the steps of scientific observation as part of a course on water body structure mapping. The aim was to find out to what extent WE can support the acquisition of a basic scientific method competence - scientific observation. For this purpose, the acquisition of factual and applied knowledge about scientific observation, motivational aspects and cognitive load were investigated. The results showed that WE promoted knowledge application, as students in the experimental groups were able to perform the steps of scientific observation more accurately. No advantage was shown by faded WE over non-faded WE. Descriptive results also revealed higher motivation and lower extraneous cognitive load in the experimental groups, although none of these differences were statistically significant. Our results add to the existing evidence that WE can be useful for developing science literacy.

Zusammenfassung

Digitale Medien besitzen ein großes Potenzial zur Unterstützung von Lernprozessen sowohl in formalen als auch in institutionellen informellen Lernorten. Die vorliegende Dissertation untersucht das Lernen mit digitalen Medien einmal in einem institutionellen informellen Kontext als systematisches Literaturreview und einmal als angewandtes Beispiel in einem formalen Setting, aber informellen Lernkontext. Um einen ersten Überblick über die Forschung zum Einsatz von digitalen Medien an institutionellen informellen Lernorten sowie Anknüpfungspunkte für weitere Forschung zu erhalten, wurde in einem ersten Schritt ein systematisches Literaturreview (Artikel A) anhand von 26 Studien erstellt. Mittels der qualitativen Inhaltsanalyse wurden die ermittelten Studien zu den allgemeinen Merkmalen der digitalen Medien und deren Funktionen sowie zu den gemessenen Ergebnissen in Bezug auf das informelle Lernen mit digitalen Medien in institutionellen informellen Lernorten untersucht. Die Ergebnisse zeigen, dass portable und stationäre digitale Medien etwa gleich häufig eingesetzt werden und oft Augmented Reality beinhalten. Die Bandbreite an möglichen Funktionen wird jedoch bei der Konzipierung noch nicht vollständig ausgeschöpft. Es hat sich gezeigt, dass digitale Medien in der Lage sind motivationale und kognitive Lernprozesse zu fördern, vor allem beim Wissenserwerb und beim Interesse sowie bei der Zusammenarbeit und sozialen Interaktion. Anknüpfend an die Erkenntnisse aus dem systematischen Literaturreview wurde als zweiter Schritt eine empirische Studie durchgeführt, die einen beispielhaften Einsatz von digitalen Medien an einem zwar formalen Lernort, aber in einem informellen Lernkontext untersucht (Artikel B), wobei der Forschungsschwerpunkt auf der Unterstützung des Lernens lag. In der Studie nutzten 62 Universitätsstudierende verschiedene Arten von double-content Worked Examples (WE), die in eine digitale Lernumgebung eingebettet waren, um die Schritte der wissenschaftlichen Beobachtung im Rahmen eines Kurses über die Gewässerkartierung zu erlernen. Ziel war es herauszufinden, inwieweit den Erwerb einer grundlegenden wissenschaftlichen Methodenkompetenz -WE der wissenschaftlichen Beobachtung - unterstützen können. Hierfür wurde der Erwerb von Fakten- und Anwendungswissen über wissenschaftliche Beobachtung, motivationale Aspekte und kognitive Belastung untersucht. Die Ergebnisse zeigten, dass WE die Wissensanwendung fördert, da die Studierenden der Versuchsgruppen die einzelnen Schritte der wissenschaftlichen Beobachtung genauer durchzuführen konnten. Keinen Vorteil zeigten faded WE gegenüber non-faded WE. Die deskriptiven Ergebnisse ergaben zudem eine höhere Motivation und eine geringere kognitive Fremdbelastung in den Versuchsgruppen, wobei keiner dieser Unterschiede statistisch signifikant war. Unsere Ergebnisse ergänzen die bestehenden Belege dafür, dass WE für den Aufbau naturwissenschaftlicher Kompetenzen nützlich sein können.

Included Publications

The present dissertation was written cumulatively and includes two English-language journal articles published in international peer-reviewed journals. The author of this dissertation is first author of both articles (Article A: 75 %, Article B: 70 %) and was primarily responsible for developing and designing the study, collecting and analyzing the data and writing the two publications.

Article A was submitted to the open-access journal Computers and Education Open in June 2021 and was accepted for publication in December 2021. The supervisor Prof. Doris Lewalter (10%) and the co-author Dr. Stephanie Moser (15%), supported the writing process by providing professional comments and proofreading the article.

Degner, M., Moser, S., & Lewalter, D. (2021). Digital Media in Institutional Informal Learning Places: A Systematic Literature Review. *Computers and Education Open, 3*, 1-11. https://doi.org/10.1016/j.caeo.2021.100068

Article B was submitted to the open-access journal Frontiers in Education in September 2023 and was accepted for publication in February 2024. The supervisor Prof. Doris Lewalter (10%) and the co-author Dr. Stephanie Moser (10%) supported the writing process by guiding the work on the data analysis, providing professional comments and proofreading the article. The two coauthors Prof. Dr. Jürgen Geist (5%) and Dr. Joachim Pander (5%) supported the writing process for article B with their professional expertise in the field of water body structure mapping.

Lechner, M., Moser, S., Pander, J., Geist, J. & Lewalter, D. (2024). Learning scientific observation with worked examples in a digital learning environment. *Frontiers in Education*, *9*, 1-10. <u>https://doi.org/10.3389/feduc.2024.1293516</u>

Contents

Acknowledgments i
Abstractii
Zusammenfassungiii
Included Publicationsiv
1 Introduction1
2 Different forms of learning (places) 4
2.1 Differentiation of the learning forms
2.1.1 Institutional informal learning places
2.1.2 Informal learning in institutional informal learning places
3 Digital media and informal learning (places) 6
3.1 Digital media and the focus on informal learning6
3.2 Functions and use of digital media in institutional informal learning places
4 Learning support through digital worked examples
4.1 The use of digitally presented worked examples to enhance learning
4.2 Scientific observation as a learning object10
5 The present research
6 Study I: Digital media in institutional informal learning places: A systematic literature review 14
6.1 Qualitative data analysis in the context of a systematic literature review
6.2 Empirical findings
7 Study II: Learning scientific observation with worked examples in a digital learning environment 18
7.1 Mixed method approach to assess scientific observation competence of university students . 18
7.2 Development of a digital learning environment with digitally presented WE for learning scientific observation
7.3 Empirical findings
8 General discussion and conclusion 23
8.1 Interpretation of empirical findings23
8.2 Limitations and suggestions for future research

8.3 Implications for research and practice	28
8.4 Conclusion	30
References	
Appendix	41
Appendix A - Script Moodle learning environment	41
Appendix B - Script Video	

1 Introduction

The definition of what characterizes a learning place has been constantly changing for many decades. The term "learning place" was first defined by the German Education Council in 1974 (Deutscher Bildungsrat, 1974). At that time, a place of learning was defined as an educational institution that is recognized within the public education system and systematically organizes learning programs. Although this term already implies a plurality of learning places, informal learning and explicit orientation towards the target group are not considered. From the 1990s onwards, an extension of the learning place concept took place with the inclusion of informal learning. In the following decades, the combination of learning places and the decentralized connection of institutions and learning places came more and more into focus (Tippelt & Reich-Claassen, 2010). Currently, in addition to formal educational institutions such as schools or universities, extracurricular and informal learning places are in the focus of the educational debate, as these learning places have a great potential for, among other things, science education (Leibniz-bildungspotenziale@dipf.de, 2023).

These learning places include the institutional informal learning places, such as museums, nature parks or botanical gardens (Schwan et al., 2014). The institutional informal learning places are characterized by the fact that they are visited individually or in groups by people of different ages and backgrounds for a variety of reasons, such as the need for education or entertainment. These learning places offer learners the opportunity to gain experiences that are sometimes not possible in a formal setting (Lin & Schunn, 2016). The authenticity of the place, for example, in zoos or aquariums allows learners to observe animals in their habitat or look at objects presented in museums in reality rather than just on pictures or videos. However, these learning experiences are characterized by often being spontaneous and uncoordinated, making the learning process mostly self-directed (Rogers, 2014) and often leading to knowledge gains based on random discoveries (Lewalter & Neubauer, 2019).

This leads necessarily to the question of how learning can be supported in these informal learning places in order to ensure learning success. Digital media seem to be appropriate for this purpose, as they contain various functions with which learning can be structured and supported. For example, information can be presented in a variety of ways, such as visually and auditorily or by mixing virtual and real environments (Ibáñez & Delgado-Kloos, 2018; Mantiri, 2014). Through the functions of adaptivity and interactivity it is also possible to make concrete reference to the learner's level of knowledge and provide a tool for active learning (Bannert & Reimann, 2009; Gerard et al., 2015; Niegemann & Heidig, 2019).

Institutional informal learning places such as museums often relate thematically to the natural sciences. With regard to the learning content to be taught at these informal learning places, they can offer the opportunity to acquire basic or general knowledge on domain-specific topics or more universal scientific methods (Lunetta et al., 2007). Due to their functions, digital media can provide appropriate support for teaching such learning content.

In addition, a specific instructional scaffold called worked examples (WE) have proven to be particularly helpful in acquiring the cognitive schemata that often have to be learned in the natural sciences (Renkl, 2014). WE follow a specific structure in that they demonstrate all the steps required to solve a task or problem. They consist of a problem statement, the solution steps and the solution itself. By focusing on the problem and solution steps while working on the task, the learner should be able to develop more generalized solutions or schemata. The learning-irrelevant, cognitive load is low with WE solutions (Sweller et al., 1998). This is because the use of WE, also in their digital form, can avoid consuming more cognitive resources than necessary to learn successive actions (Renkl, 2017). The benefit of using WE in science is well established by studies (Barbieri et al., 2021; Booth et al., 2015). However, only a few studies specifically address the impact of WE on the acquisition of basic science competencies involving heuristic problem-solving processes (Hefter et al., 2014; Koenen et al., 2017; Schworm & Renkl, 2007).

The overall aim of this dissertation is to show how (mobile) digital media are used in informal learning contexts and to what extent the media can support the informal learning in terms of cognitive and motivational aspects. As this is still a rather new field of research with just older and non-systematically analyzed literature reviews (Hawkey, 2004; Sefton-Green, 2004), the first step was to review the current status quo of the research field by conducting a systematic literature review in order to get a first impression of how (mobile) digital media are used to support informal learning with a focus on institutional informal learning places (Article A). Here, the focus of the evaluation was on the general characteristics, learning-relevant functions and learning outcomes for the digital media examined in institutional informal learning places. The term institutional informal learning places was chosen for the study because the focus should be explicitly on informal learning in specific institutions. In the systematic literature review, the focus is on the use of digital media during the individual informal learning learning process of visitors in the sense of a freely selectable learning activity in the respective setting.

After attempting to answer the overall goal by means of qualitative research, the second step was to develop a mobile digital learning environment based on the findings of the literature review. In addition, the self-developed learning environment should be designed to be motivating and not cognitively overloaded and providing factual knowledge and applied knowledge (Article B).

2

Specifically, the study examined the effects of different forms of digitally presented WE (no WE vs. non-faded WE vs. faded WE) using learning videos on students' cognitive and motivational outcomes. The WE are heuristic double-content We, which include two learning domains (learning domain and exemplifying domain). The insights gained from this research provide a practical extension of the findings from Article A in order to give hands-on recommendations for this research field. Furthermore, the combined perspective of factual and applied knowledge as well as motivational and cognitive aspects provides another value of the study. Both articles have been published in peer-reviewed international journals.

At the beginning of the thesis, chapter 2 provides an overview of different learning places with a special focus on institutional informal learning places. In chapter 3, the use of digital media to support informal learning in these learning places is presented. How specific learning support in the form of digital worked examples can look like, is explained in chapter 4. Following the theoretical part, chapter 5 presents the two studies with their specific research questions. In Chapter 6, Study I is presented with regard to the research design and the results. Chapter 7 presents Study II, including the research design and results, as well as a description of the digital learning environment. The chapter 8 contains the overall interpretation and discussion of the results as well as the practical implications. At the end, the limitations of the studies are mentioned and recommendations for future research are given. The thesis closes with the conclusion.

2 Different forms of learning (places)

2.1 Differentiation of the learning forms

In Study I, the focus was on the investigation of learning in institutional informal learning places in order to examine the use of digital media during the individual informal learning process of visitors, in the sense of a freely selectable learning activity in the respective setting. However, it is important to understand and differentiate between the different forms of learning in other learning places, as they each encompass different learning contexts and methods. Knowledge of these different learning contexts and methods and methods can help to develop effective learning strategies.

A general distinction is made between the three forms of formal, non-formal and informal learning (Lewalter & Neubauer, 2019). Formal learning traditionally takes place in educational or training institutions and occurs under pedagogical guidance. Learning in such educational institutions is structured in terms of learning content, learning objective, learning time and learning method. The content to be taught is based on the curriculum and learning leads to formal certification. From the learner's perspective, formal learning is mandatory, goal-orientated and intentional. In contrast, non-formal learning is typically not offered by an educational or training institution, but takes place at specific locations, such as in-house training. Also, in this case learning is systematically structured in terms of learner's perspective, non-formal learning is voluntary, goal-orientated and intentional. Informal learning, on the other hand, often takes place in everyday life in a non-staged setting. It is not structured in terms of learning objectives, learning time or learning time or learning support and does not lead to certification. From the learner's perspective, non-formal learning is voluntary, goal-orientated and intentional. Informal learning, on the other hand, often takes place in everyday life in a non-staged setting. It is not structured in terms of learning objectives, learning time or learning can be intentional from the learner's perspective, but does not have to be (European Commission, 2002; Lewalter & Neubauer, 2019).

2.1.1 Institutional informal learning places

Institutional informal learning places such as museums, zoos, aquariums or science centers represent a specific learning setting that enables learners to learn independently and, in an action-oriented way. They differ from formal educational settings such as schools in that learners can have a variety of experiences there that are not possible in formal settings (Lin & Schunn, 2016). Another characteristic of these learning places is that they are visited by a very heterogeneous audience in terms of age and preconditions (Schwan et al., 2014) and when visiting the places as a group or individually, the goals and intentions of the visit can be very different (Falk, 2009). These learning places are often visited because of the need for entertainment, education, aesthetic appreciation, recreation, social exchange or because of the place itself (Schwan et al., 2014). In these learning places, information are often presented through real and authentic objects such as artifacts or live animals and supplemented by hands-on activities and narratives in this setting (Schwan et al., 2014). This type of learning place often consists of a large room that has many focal points of attention and presents information simultaneously throughout the room.

2.1.2 Informal learning in institutional informal learning places

As shown in 2.1.1, there are different types of institutional informal learning places in which learning takes place in a very specific way. Learning in institutional informal learning places is lifelong and selfdirected learning, which usually takes place without direct guidance (Schugurensky, 2000). This means that it is not structured in terms of learning goals, learning time or learning support. Instead, learning is problem-solving orientated, whereby the learning material is predominantly viewed holistically (Eshach, 2007; Lewalter & Neubauer, 2020). On the one hand, this means that learners initiate and plan the informal learning process themselves and carry it out purposefully and intentionally. However, this does not necessarily have to be the case. The learning process in institutional informal learning places can be self-initiated, but not purposeful and casually (Lewalter & Neubauer, 2019). The learning outcomes achieved in this manner are primarily based on experiential knowledge, which means that informal learning does not lead to certification (Rogers, 2014). It has also been shown that informal learning activities can enhance scientific thinking skills (Gerber et al., 2001). The experiences gained through informal learning foster learning and interest in such a way that it can lead to ongoing engagement in science activities (Morris et al., 2019).

3 Digital media and informal learning (places)

3.1 Digital media and the focus on informal learning

Due to rapid technological progress, digital media can be found in many areas of life. Especially for the younger generation, digital media have become an integral part of life (Mantiri, 2014). This is accompanied by the diverse use of digital media, whereby the support of learning processes plays a major role. (Mobile) digital media can be defined as electronic devices on which information can be saved and transferred in digital form. In addition to stationary digital media such as tabletops or computer, mobile web-based media such as smartphones or tablets are primarily used to support learning in order to access (learning) information independently of time and place. The retrieved (learning) information can then be presented in different ways via digital media, for example, as texts, images, sounds, videos, animations or a combination of these elements (Mantiri, 2014). This creates the opportunity for digital learning material to be designed in such a way that it can be used by learners according to their own preferences, prior knowledge or learning pace. In terms of "ubiquitous learning", mobile digital media thus offer the opportunity to design learning processes in a flexible and self-directed way (Moser, 2017).

However, it must be considered here that, especially in self-directed learning environments as is often the case in informal learning places, the use of digital information in itself does not automatically lead to an enhancement of the learning process. For this reason, further supportive measures are necessary to guide the learning process in order to really support learning processes using digital information (Bannert & Reimann, 2009; Lin et al., 2017; Moser et al., 2017). On a cognitive and motivational level, these can be, n, design features that enable interactivity and adaptivity (Bannert & Reimann, 2009; Gerard et al., 2015; Niegemann & Heidig, 2019). Studies have already shown that digital media appear to have a positive influence on the learning process for learner's subject knowledge (Chien, 2012), motivation (Lin et al., 2017) and media literacy (Ungerer, 2016). The use of digital media to support learning can be especially appropriate in institutional informal learning places (Schwan et al., 2018). Here, digital media offer an exciting, interactive and educational experience and thus combine leisure entertainment and education (Schwan, 2015). In museums, for example, information is prepared with the help of digital media in order to explore exhibits in a self-directed manner instead of passively viewing objects (Schwan, 2015). New digital applications such as AR options can be implemented here, for example, which combine the physical and virtual worlds of the learners by expanding the real world of the learners with additional virtual information on a display (Ibáñez & Delgado-Kloos, 2018).

This results in two roles for digital media: In one case, digital media can be used to complement the learning experience by presenting a complex principle using animation on a screen. Secondly, the digital medium itself can provide authentic insights, for example, when learners actively use the medium to acquire new knowledge on their own. Digital media are usually found in museums in their complementary role (Kampschulte et al., 2019).

3.2 Functions and use of digital media in institutional informal learning places

Digital media offer the advantage that they can provide various functions, adapted to the respective learning situation. Three didactically relevant functions in teaching-learning processes can be considered as follows (Ojstersek & Kerres, 2010): 1) Digital media as knowledge tools for the presentation of information, in order to present difficult-to-understand facts more clearly through different presentations. 2) Digital media for communication and cooperation to support collaboration and communication in synchronous and asynchronous form between individuals and groups. 3) Digital media for controlling the learning process by using digital learning programs to enable the presentation of learning content in the ongoing learning process or self-directed processing of learning content.

An extension to five functions of digital media is made by Petko (2014): 1) Digital media have here the function as information and presentation means, in order to represent learning contents understandably as well as to present the information in different way (text, audio, video). 2) Digital media can be used to make learning tasks more diverse, for example, by linking various multimedia materials with reality. 3) Digital media as a tool and work equipment offer the possibility to support the work by word processing and presentation programs. 4) Digital media for learning support and communication can promote interaction, for example, through chats, forums, audio and video conferences. 5) Finally, digital media for test evaluation offer the possibility of conducting exams more efficiently, evaluating them automatically and reporting back.

Based on Petko's five functions, Kampschulte et al. (2019) analyzed the functions of the media used in 120 informal learning places. In fact, the results show that digital media are primarily used as "information and presentation tools" in informal learning places. In second and third place are media for "designing learning tasks" and as "tools and work equipment". The media are rarely used for "learning guidance and communication" and are only used for "examination and assessment" at one of the learning places.

4 Learning support through digital worked examples

4.1 The use of digitally presented worked examples to enhance learning

Especially in the fields of STEM education (science, technology, engineering and mathematic), it is often necessary to learn certain laws or principles. This often concerns the acquisition of domain-specific topics such as the application of a mathematical multiplication rule or more general scientific methods such as conducting the steps of a scientific observation (Lunetta et al., 2007). Digital media can be used here to support STEM learning, such as through the use of digitally presented WE (Booth et al., 2015; Renkl, 2014). The digitally presented WE enables the content to be presented in different ways, for example, via video, audio, text or images or in a combination of different formats. This is beneficial because it allows the content to be adapted to the needs of the learners, so that it can be used individually according to their own prior knowledge or learning speed (Mayer, 2001). WE are designed to consist of a problem statement, the solution steps, and the solution itself (Atkinson et al., 2000; Renkl, 2014; Renkl et al., 2002). This information can be provided to learners in a timely, motivating, and engaging manner using digital media to make learning effective and on-demand (Mayer, 2001).

Studies show that digitally presented WE leads to higher learner satisfaction, positive attitudes and content knowledge (Dart et al., 2020). The use of digital WE can demonstrate positive effects on self-directed learning in terms of learning location, learning time, and learning speed (Kay & Edwards, 2012). The addition of self-explanatory prompts to digital WE have been shown to be superior to learning methods such as hypertext learning and observational learning (Berthold et al., 2009). The successful use of WE are significantly dependent on the influence on the cognitive load during learning, which is based on the assumption of the Cognitive Load Theory (Sweller, 2006).

The Cognitive Load Theory analyses the cognitive load when acquiring knowledge. The basic idea of the theory is that the cognitive capacity of the human working memory is limited, so that learning is hindered if a learning task requires too much capacity. The extent of the cognitive load during learning depends on how complex the learning content is and how the learning materials are designed (Skulmowski & Man Xu, 2021; Sweller, 2006). If the cognitive load is too high for the capacity of the working memory, this leads to cognitive overload, which has a negative impact on learning success. A differentiation is made between three forms of cognitive load (Sweller, 2006). Intrinsic cognitive load is caused by the complexity and difficulty of the task itself. The more complex and difficult a task is, the higher the intrinsic cognitive load can be. The extraneous cognitive load is due to the design and presentation of the learning material and the learning environment. A high extraneous cognitive load can lead to reduced performance on a task.

Germane cognitive load refers to the mental effort required to understand and learn new concepts and information. Germane cognitive load is important to the learning process as it helps to ensure that information can be effectively processed and remembered (Sweller, 2010). Here, studies show that the use of WE can lead to a decrease in extraneous load, while leading to an increase in germane load (Paas et al., 2003; Renkl, 2014). Regarding intrinsic load, it is still not conclusive whether WE have an influence on this (Gerjets et al., 2004).

Furthermore, WE have been shown to be particularly effective with novices in terms of learning gain (Bokosmaty et al., 2015; Sweller et al., 1998; Van Gog et al., 2011), but can lose this advantage as knowledge gain progresses due to the expertise reversal effect. The expertise reversal effect means that the effectiveness of the instructional material is strongly dependent on the learner's level of knowledge. While inexperienced learners may benefit from more guidance, experienced learners may have a load on their working memory, because some or all of the instructions may be redundant for them (Kalyuga et al., 2003). This phenomenon can be prevented with so-called scaffolding learning, in which so-called faded WE are used to gradually fade out solution steps as learners' knowledge and expertise grow (Renkl, 2014). Here, faded WE, in contrast to non-faded WE have been shown to improve near knowledge transfer while reducing errors (Renkl et al., 2000).

In terms of learning motivation, the reduction of intrinsic and extraneous cognitive load through WE show a positive impact (Van Harsel et al., 2019). In particular, positive effects have been demonstrated on interest (Gupta, 2019; Van Gog & Paas, 2006) and with regard to learner satisfaction and emotions (Um et al., 2012). Again, fading and non-fading WE have different effects on learners. Fading WE are more motivating for more experienced learners, while non-fading WE can be particularly motivating for learners without prior knowledge (Paas et al., 2005). In addition, it can be further assumed that learning environments that use WE may have an impact on learners' situational interest and basic needs. This assumption is based on the fact that learning environments are influenced by motivational aspects in the learning process, such as situational interest (Lewalter & Knogler, 2014), motivationally relevant experiences, such as basic needs, or by motivational attributes of the learning process, such as self-determined motivation (Deci & Ryan, 2012).

Situational interest is a content-related motivational quality that develops in a current learning situation and is linked to it (Lewalter & Willems, 2009). It is based on the situational characteristics of the learning environment, the individually perceived interest of the learning material and the emotional experience during the learning process. Situational interest is divided into two phases, the catch phase ("triggered interest") and the hold phase ("maintained interest") (Neubauer, 2015). The catch phase refers to the first occurrence of situational interest, where one's own attention is drawn to an object and interest is awakened.

9

In the following hold phase, interest in the learning situation should be maintained, for example, by the learner seeing the learning object as personally relevant and therefore wanting to become more involved with it (Lewalter & Willems, 2009; Neubauer, 2015). There is a close connection here to the value-based component of basic scientific literacy, which emphasizes the appreciation of science and scientific research, as well as the willingness to engage with it. This confirms the appropriateness of situational interest as a motivational indicator for measuring basic scientific literacy (Neubauer, 2015). Learning environments that foster autonomy, competence and social integration in the sense of basic needs can contribute to the development of situational interest in particular (Lewalter & Willems, 2009). The need for autonomy is expressed in a person's efforts to experience themselves as acting independently, for example, by defining the goals and procedures of their own actions for themselves. The experience of competence refers to a person's need to experience themselves as capable of acting in the face of challenges in learning and work situations and to be able to manage tasks under their own power. The need for social integration is based on a person's general desire for social contact and is expressed in the effort to achieve social acceptance in the relevant peer group.

Furthermore, with regard to WE, a distinction can be made between algorithmic and heuristic WE as well as WE with one and two contents (Koenen et al., 2017; Reiss et al., 2008; Renkl, 2017). As the focus of the present study is on heuristic WE with two contents, only these are described below. According to Koenen et al. (2017), heuristic WE used in STEM fields, mainly include basic scientific working methods, such as conducting experiments. Double-content WE include two learning domains relevant to the learning process (Renkl et al., 2009). Here, the first learning domain contains the process or concept to be learned, in this study the method of scientific observation. The second exemplifying domain contains the specific content on which the process or concept from the learning domain is learnt. In this study, this is the mapping of a water body structure.

4.2 Scientific observation as a learning object

As mentioned above, learning in science is often about the acquisition of fundamental principles related to, for example, universal scientific methods (Lunetta et al., 2007). An example is learning the steps of scientific observation, which is fundamental to all scientific activities and disciplines (Kohlhauf et al., 2011). In contrast to unspecific everyday observation, which merely involves the noticing and description of certain characteristics (Chinn & Malhotra, 2001), scientific observation is a complex activity for the method of knowledge acquisition. It is defined as theory-based, systematic and selective observation of concrete systems and processes without basic manipulation (Wellnitz & Mayer, 2013).

Similar to experimentation, scientific observation is also about developing an understanding of the criteria for appropriate experimental design in relation to the development, measurement and evaluation of outcomes (Brownell et al, 2014; Dasgupta, Anderson & Pelaez, 2014; Deane et al, 2014; Sirum & Humburg, 2011).

In detail, scientific observation consists of the six steps of 1) formulating the research question(s), 2) deriving the null hypothesis and alternative hypothesis, 3) planning the research design, 4) conducting the observation, 5) analyzing the data and 6) answering the research question(s) (Wellnitz & Mayer, 2013).

The WE described in the chapter 4.1 are shown here to be suitable to support the learning of scientific observation (Booth et al., 2015; Renkl, 2014). This didactic support on a cognitive level by means of digitally presented WE is important so that observation can be carried out in a clear and structured manner (Eberbach & Crowley, 2009; Ford, 2005). This is because, especially in informal learning contexts, scientific observation often takes place in a spontaneous and uncoordinated manner and knowledge gain is due to random discoveries (Jensen, 2014).

5 The present research

Against the background of the growing importance use of digital media in school and university educational contexts, this dissertation contributes to providing on the one hand, an up to date and systematic overview of the research field of the use of digital media in institutional informal learning places to support informal learning and on the other hand to analyses the effectiveness of a self-developed digital learning environment in relation to double-content WE in an informal learning context. The results of the studies presented in the two articles of the dissertation contribute to finding out how digital media should be designed in highly self-regulated educational contexts in order to be as motivating and cognitively less loaded as possible, as well as to contribute to maximizing knowledge acquisition.

Article A (Digital Media in Institutional Informal Learning Places: A Systematic Literature Review) Learning contexts and educational contexts are becoming increasingly diverse. In addition to the classic formal learning places such as school or university, institutional informal learning places such as museums, botanical gardens or zoos become more and more important (Schwan et al., 2014). However, learning in precisely such learning places is often confronted with the difficulties that the learning process here often takes place spontaneously and uncoordinated (Rogers, 2014). For this reason, it is important to provide support, for example, through the use of digital media. Because of their different functions, such as the possibility to present information in a variety of ways, they have a high potential to structure informal learning. The use of digital media to support learning in institutional informal learning contexts is a field of educational research with rather older and not systematically examined literature reviews. And even in the more recent literature, mainly applicationspecific analyses can be found, for example, review articles on AR applications. To address this research gap, an updated, comprehensive, and systematic literature review was conducted to investigate research findings related to the use of digital media in institutional informal learning places. The research focused primarily on the general characteristics, learning-relevant functions and learning outcomes of using digital media in this educational context.

This led to the following three research questions:

RQ1: What are the general characteristics of the examined digital media in institutional informal learning places?

RQ2: What are the functions of the examined digital media in institutional informal learning places that are relevant for learning?

RQ3: What learning outcomes were measured as they relate to informal learning with digital media in institutional informal learning places in the examined studies?

Article B (Learning Scientific Observation with Worked Examples in a Digital Learning Environment) Based on the findings the systematic literature review, the aim of Study II was to develop and analysis a digital learning environment and to test the practical application in an informal learning context with regard to knowledge acquisition, motivation and cognitive load. While knowledge acquisition and motivation were already investigated aspects in the systematic literature review, the cognitive load was not investigated in detail. However, especially with the use of WE, the cognitive load is an important aspect to be investigated, as WE can support the development of cognitive schemas during learning in order to avoid an overload of cognitive resources during learning (Renkl et al., 2004; Renkl, 2017). Previous studies showed that faded WE performed better compared to non-faded WE in terms of knowledge acquisition and cognitive overload (Renkl et al., 2000). The study was designed in the context of a university water body structure mapping workshop in which during a field trip several river sections of a small Bavarian river were mapped. Three different types of double-content WE (no WE, faded WE, non-faded WE) in the form of learning videos were investigated to what extent they can support the acquisition of a basic scientific methodological competence - conducting scientific observations. So, the focus is on learning scientific observation (learning domain) through river structure mapping (exemplifying domain), which takes place with the support of digital media in a formal (university) setting but in an informal learning context (field trip in nature).

This led to the following three research questions:

RQ1: To what extent exist differences between the three conditions (no WE, faded WE, non-faded WE) with regard to learners' competence acquisition (acquisition of factual knowledge about the scientific observation method (quantitative data) and practical application of the scientific observation method (quantified qualitative data))?

RQ2: To what extent exist differences between the three conditions with regard to learners' motivation (situational interest and basic needs)?

RQ3: To what extent exist differences between the three conditions with regard to cognitive load?

It is assumed that (Hypothesis 1), the integration of WE (faded and non-faded) leads to significantly higher competence acquisition (factual and applied knowledge), significantly higher motivation and significantly lower extraneous cognitive load as well as higher germane cognitive load during the learning process compared to a learning environment without WE. No differences between the conditions are expected regarding intrinsic cognitive load. Further, it is assumed (Hypothesis 2) that the integration of faded WE leads to significantly higher competence acquisition, significantly higher motivation and lower extraneous cognitive load as well as higher germane cognitive load during the learning processes compared to non-faded WE. No differences between the conditions are expected with regard to intrinsic cognitive load.

6 Study I: Digital media in institutional informal learning places: A systematic literature review

6.1 Qualitative data analysis in the context of a systematic literature review

For the systematic literature review as Study I, a literature search was conducted between November 2020 and February 2021 according to PRISMA guidelines (Higgins et al., 2019; Siddaway et al., 2019). After determining inclusion and exclusion criteria, established online research databases on education (Scopus and FIS (include ERIC, EBSCOhost ebooks and BASE)) were searched for studies. Search terms covered the three main topics of the research questions were used for this purpose: Media, Informal Learning and Institutional Informal Learning Places. The search of the databases yielded a total of 125 hits, 52 from the FIS database and 73 from Scopus. Four duplicates were removed from these hits. Subsequently, the titles and abstracts of the remaining 121 hits were reviewed, resulting in the exclusion of 92 hits. The remaining 29 hits were now reviewed in detail and another 19 hits were excluded. A total of 10 matching studies were thus identified. In the next step, a backward as well as a forward search was performed on the 10 studies and the excluded literature reviews to find additional relevant studies. Through these procedures, additional 16 studies were found. In total, the systematic literature review thus examined 26 studies. These 26 studies are marked with an asterisk in the references list. A more detailed description of the literature search including the exact search syntax can be found in the article in chapter 3.1. The studies were evaluated using the method of qualitative content analysis (Aspers & Corte, 2019; Mayring, 2015). For this purpose, corresponding categories were formed for each of the three research questions in order to group the studies according to their common characteristics.

The first category examined the hardware and software to determine the general characteristics of digital media. The category "hardware" included the subcategory "device" (medium used) and the subcategories "mobile medium" and "stationary medium". The category "Software" included the subcategories "Accessibility", "Name of the software used" and "Digital format". The second category examined six didactic functions and one sub-function following Kerres (2000) and Petko (2014) to determine the general media functions. Here the categories "information" (only availability of information), "task" (giving an instruction to work), the subcategory "processed task" (processing the task in the real or the virtual world), "documentation and processing" (creating photos, audio recordings or notes) "cooperation or collaboration" (instruction to work together to process a task), "communication" (possibility to contact other users), "evaluation and feedback" (receiving feedback) were examined. The third category examined the measured learning outcomes. For this, the five categories "research approach," "data collection," "cognitive outcomes," "motivational outcomes," and "outcomes related to collaboration and social interaction" were examined.

The coding of the 26 studies was done manually separately according to their characteristics and then classified according to the established categories. To check the coding, eight manuscripts were then randomly selected and cross-coded by an external coder. A total of 208 cases were analyzed, with 17 cases not matching. The analysis revealed a high to very high degree of agreement (Cohen's $\kappa = 0.763$ to Cohen's $\kappa = 0.910$). It can be assumed that the category system is clearly formulated and that the units of analysis can be correctly assigned. With regard to the 17 differently coded cases, both coders exchanged views and adapted the coding guide on the basis of consensus.

6.2 Empirical findings

Before the results are presented, structured according to the research questions, a short overview of the studies analyzed is given in order to be able to better classify the further results. The studies analyzed are predominantly from the natural sciences (biology and physics) and rarely from the fields of art, history or music. The institutional informal learning places found in the articles are primarily museums, especially natural science museums. Since 2016, significantly more studies have been published and the institutional informal learning places have become more heterogeneous and, in addition to well-known learning places such as science and nature centers, summer camps and nature parks have been found (Zimmermann et al., 2014; Zimmermann et al., 2016; Fränkel et al, 2020; Knoblich, 2020). With regard to the duration of the visit, it was revealed that visitors spend an average of 30 to 60 minutes at the learning place and the visit mainly takes place in groups of two or three.

With regard to the first research question, which examines the characteristics of digital media, the results showed that stationary digital media are used approximately as frequently as mobile digital media. However, since 2018 mobile digital media such as smartphones or tablets have been increasingly used, although the use of smartphones is very rare in contrast to tablets. One explanation for this could be that, due to rapid technological progress, smartphones are now better equipped and can be used to play complex applications (Khaddage et al., 2016). Most visitors now bring their own smartphones with them so to use according to the "bring your own device (BYOD)" approach. Applications are developed specifically for the studies with approximately the same frequency as using existing applications. However, self-developed applications have been increasingly implemented since 2018. AR visualizations have become quite common in recent years, but they differ significantly in terms of complexity. While complex AR visualizations such as 2D and 3D avatars have been used since 2019 (Moorhouse et al., 2019; Hammady et al., 2020), AR was mostly used as dynamic visualizations or instructional labels in previous studies (Yoon, et al., 2012a; Yoon, et al., 2012b; Wang et al., 2012).

With regard to the second research question, which examines the functions of digital media, the evaluation showed that the most frequently used function is the provision of information and tasks. This corresponds to the findings of Kampschulte et al. (2019), where media are used in particular for information and presentation purposes as well as for the design of learning tasks. In the studies analyzed, the tasks are mainly completed in the virtual world, with different levels of complexity. In addition to short work instructions ("try to complete the cycle", Yoon et al., 2012b), there are also detailed instructions on how to explore the museum (Clayphan et al., 2018). Both in this evaluation and in the study by Kampschulte et al. (2019), the documentation and editing function comes in third place, which mostly includes tools for taking photos or saving notes. The existence of a communication function in the media tool, for example, in the form of chats or emails, is very rare both in the present study (Evans et al., 2014; Zimmermann et al., 2014) and in the study of Kampschulte et al. (2019). Overall, an increase in the number of digital media functions can be seen from 2018 onwards.

With regard to the third research question, it has been shown that qualitative or mixed-methods are primarily used for data collection. One reason for this could be the relatively new field of research, where the focus is on obtaining comprehensive results (Mohajan, 2019). Aspects such as engagement or social interaction can be measured more accurately through qualitative data collection, for example, by means of observation (Apers & Corte, 2019). Quantitative research is less common and is mainly used to evaluate the acquisition of knowledge. Most studies use a research design without a control group to investigate different conditions in the media tool. Also, the use of media is rarely tested against no use of media, if then in relation to the acquisition of knowledge. Especially at the beginning of media research, the investigation of the effectiveness of digital media with learning without digital media was prioritized, even if a specific analysis of the conditions was often neglected (Ammad-ud-din et al., 2014).

With regard to the measured outcomes, the results show that the outcomes can be divided into three categories: cognitive outcomes, motivational outcomes and outcomes relating to collaboration and interaction. With regard to the cognitive outcomes, the analysis found that mainly content knowledge and conceptual knowledge were measured. Other cognitive outcomes measured, but very rarely found, were cognitive understanding, preferred learning style, cognitive theorizing skills, media literacy, procedural knowledge and digital competence. Only positive cognitive outcomes related to the use of digital media were reported in all studies analyzed. With regard to the motivational outcomes the analysis showed that different motivation qualities, interest, and visitor engagement are measured. Regarding motivation, different aspects are assessed: general motivation, intrinsic motivation, internal motivation and external motivation. In addition, interest is partly stated more specifically: situational interest, general interest and science interest.

In all studies reviewed, only positive motivational outcomes are reported. With regard to visitor engagement, the studies analyzed showed that interactive tabletops have a positive impact on visitor engagement. One study (Dieck et al., 2016) reported negative effects on visitor engagement and social acceptance of visitors due to the use of wearable AR applications. In terms of collaboration and interaction, the studies analyzed showed that social interaction, collaboration and family conversations were measured. In all studies analyzed, collaboration and interaction were evaluated positively. A more detailed description of the results including the qualitative analysis can be found in the article in chapter 4.

Overall, the results show that institutional informal learning places have become more and more diverse in recent years and that, in addition to museums, galleries or nature parks are increasingly coming into focus. It also shows that the design of digital media tools is becoming more and more complex, for example through the increased use of 3D animations, and that portable digital media are preferred. Not nearly all the functions that digital media can provide are currently used, but mostly the provision of information and tasks. Furthermore, the results suggest that digital media promote and support motivational and cognitive informal learning processes, especially in knowledge acquisition and interest as well as collaboration and social interaction.

In summary, this study has broadened research on the still very new field of educational research on the use of digital media in institutional informal learning places by systematically correlating the individual studies for the first time in a systematic literature review. The study thus provides important insights for media-based educational research. The overarching conclusion of the study is that digital media at institutional informal learning places offer appropriate support possibilities to enhance informal learning.

7 Study II: Learning scientific observation with worked examples in a digital learning environment

7.1 Mixed method approach to assess scientific observation competence of university students

In Study II, 62 science students and doctoral students from a German university participated (age M = 24.03 years; SD = 4.20; 36 females; 26 male). The participants had different levels of prior knowledge of scientific observation (N=37, intermediate level knowledge) and water body structure mapping (N=7, intermediate level knowledge). N=25 had no experience in both. Two participants were excluded retrospectively due to lack of posttest. The study was conducted as a field experiment with a 1factorial, guasi-experimental, comparative research design (pretest and posttest) where participants were randomly assigned to one of three conditions. Two online questionnaires (pretest and posttest) were used to collect the data. The pretest included socio-demographic questions, questions about the course of study, number of semesters and questions about prior knowledge about scientific observation. Also, the independent preparation based on the learning materials for the water body structure mapping was questioned (Bayerisches Landesamt für Umwelt, 2019; Walker et al. 2007). Furthermore, the pretest included a factual knowledge test on scientific observation and questions for calculation the self-determination index. The factual knowledge test for the quantitative assessment of scientific observation competence consisted of 12 single-choice questions. The questions are based on Wahser (2008, adapted from Koenen, 2014) and are adapted to scientific observation. For the calculation of the self-determination index (SDI-index), Thomas and Müller's (2011) scale for selfdetermination was used in the pretest.

In the posttest, the same knowledge test was used for comparison. In addition, questions were asked about basic needs, situational interest, measures of cognitive load and questions about the usefulness of the WE. The basic needs were measured with the scale by Willems and Lewalter (2011). Situational interest was measured with the scale by Lewalter and Knogler (2014; Knogler et al., 2015; Lewalter, 2020). The intrinsic cognitive load and the extraneous cognitive load were measured with the scales from Klepsch et al. (2017). The germane cognitive load was measured with the scale from Leppink et al. (2013). The usefulness of the WE were measured with the scale from Renkl (2001) and with the scale from Wachsmuth (2020). All scales used in the study are proven instruments for measuring the results that fulfil the quality criteria of objectivity, reliability and validity. Furthermore, applied knowledge in the form of answers written by the participants in the digital learning environment (Chapter 7.2) was evaluated by using a self-developed coding scheme. A detailed description of the coding scheme can be found in the article in chapter 3.4.1.

Using the coding scheme, 600 cases to be coded (60 participants with 10 responses each) were identified and coded by the first author. Of these cases, 240 cases (24 participants, 8 from each course) were randomly selected and cross-coded by an external coder for verification. The assessment agreed in 206 of the coded cases. The cases where the raters disagreed were discussed together and a solution was found. A Cohen's κ = 0.858 results from this, indicating a high to very high level of agreement. This suggests that the category system is clearly formulated and that the individual units of analysis could be correctly categorized.

To answer the research questions analyses were conducted using SPSS version 28. Descriptive statistics (means, standard deviation) were calculated for the variables. Using ANOVA, there were no significant differences in the variable's prior knowledge and SDI index between the three groups, which is why they were excluded as covariates. A single factorial analysis of variance (ANOVA) with repeated measures was used to compare the three groups (no WE vs. non-faded WE vs. faded WE) in terms of the increase in factual knowledge about the scientific observation method from the pretest to the posttest. Furthermore, a multivariate analysis (MANOVA) was calculated with the three groups (no WE vs. non-faded WE vs. faded WE) as the fixed factor and the dependent variables practical application of the scientific observation method, situational interest, basic needs, and cognitive load. Bonferroni-adjusted post-hoc analyses were calculated to determine differences in applied knowledge among the three groups.

7.2 Development of a digital learning environment with digitally presented WE for learning scientific observation

When developing the digital learning environment, the focus was on acquiring the competence of scientific observation. With the help of the digital learning environment the six steps of scientific observation, the formulation of research questions and hypotheses, the planning and implementation as well as the analysis and interpretation of the data, were explained to the students so that they are able to conduct a scientific observation independently at the end of the digital learning environment. In order to be able to learn the concept of scientific observation, a second learning domain is needed, through which the concept is learned (Koenen, 2014; Renkl et al., 2009). The second learning domain focused on the mapping of a water body structure, which was to be used to practice scientific observation. Learning how to conduct a water body structure mapping thus represents the second learning aim, which should be taught using the digital learning environment, but was not evaluated as part of the study.

The Moodle software was used for the technical realization of the digital learning environment. A specific layout was designed, which is structured like a book in which it is not possible to turn back. This is important insofar as the participants do not have the possibility to revisit information in order to keep the conditions comparable as well as distinguishable. The learning videos that were added to the learning environment were recorded with Power Point, set to audio and converted into a video file. The structure of the learning environment is described below, more detailed information on the scripts for the learning environment and the videos can be found in the article and in the appendix A and appendix B.

The digital learning environment started with a short welcome and an introduction to the course navigation followed by the pretest. A short hypothetical scenario was then described, which was the framework for the course. Participants were told that they were employees of an urban planning office who were tasked with analyzing the ecomorphological condition of a small river near a Bavarian town. The river consists of five sections, each of which has to be mapped separately. After this hypothetical framing, the mapping for the first section began. At the beginning of each section, a research question and corresponding hypotheses on the ecomorphological condition of the river section had to be formulated and written down. Afterwards, data on the river section was collected using a paper-based mapping sheet and afterward the data was then analyzed. At the end, the research question and hypotheses were answered according to the evaluation. This structure was maintained for the mapping of all five sections.

As already mentioned, the digital learning environment was designed with Moodle. In Moodle there were three courses that were identical in structure and design, the only difference was the implementation of the WE in the form of videos. The students were randomly assigned to one of the three courses, so that each course consisted of 20 participants. The course that served as the control group did not include WE videos to support the scientific observation method. The course contained only short written work constructions. The course with the non-faded WE included three videos for each of the five sections. Each of the three videos showed two steps of scientific observation, so that all six steps of scientific observation are explained in one section. The first video on research questions and hypotheses explained what constitutes scientifically based research questions and hypotheses. A practical example was given with explanations and tips on how to formulate research questions and hypotheses. The second video explained the two steps of planning and implementation. A practical example was given with explanations and tips for planning and implementing the scientific observation. The third video explained how to analyze and interpret the collected data. For this purpose, a practical example was given with explanations and tips on how to analyze and interpret data. After these three videos for mapping the first water body section, the mapping of the next section with the same structure began.

20

The course with the faded WE has the same number of videos as the course with the non-faded WE. Here, the information in the videos is gradually reduced. In the first section, all three videos are available in complete form. In the second section the information have been reduced by omitting the tip in all three videos. In the third section the tip and the practical example have been omitted. In the fourth and fifth sections there were only the written work instructions without videos.

7.3 Empirical findings

In the following, the results of the study on the use of WE to support scientific observation of students are explained according to the research questions.

The first research question related to the students' increase in knowledge. The results indicated that there were no significant differences in the factual knowledge test results between all three groups, and specifically between the two experimental groups. These findings contradict existing literature, which suggests that WE have a positive impact on knowledge acquisition (Renkl, 2014). Furthermore, research suggests that faded WE are more effective for knowledge acquisition and transfer compared to non-faded examples (Renkl, 2014; Renkl et al., 2000).

However, the study results revealed major significant differences when analyzing applied knowledge. Here, the increase in applied knowledge when conducting scientific observation steps was significantly lower for the participants in the control group than for the participants in the two experimental groups. The group that had the non-faded WE in the digital learning environment performed better than the group with the faded WE. Since no significant differences were found between the two experimental groups, it can be assumed that both faded and non-faded WE are appropriate for teaching applied knowledge about scientific observation in the learning domain (Koenen, 2014). But here too, the results of the study are different from the existing literature. According to this, the faded version results in the highest knowledge transfer (Renkl et al., 2000). Nevertheless, the results of Renkl (2014) and Hesser and Gregory (2015) were confirmed, according to which the faded version of the WE are also appropriate for supporting learning, even if the non-faded WE performed best.

The second research question examined the participants' motivation in relation to situational interest and basic needs. The results showed no significant differences across all three groups or between the two experimental groups. When evaluating the descriptive results, a slightly higher level of motivation (situational interest and basic needs) was found in the two experimental groups compared to the control group. The descriptive results confirm the existing literature to the effect that WE lead to higher motivation relevant to learning (Paas et al., 2005; Van Harsel et al., 2019). Further data analysis showed that both test groups consider the benefits of WE to be high and rate WE positively. The third research question examined the cognitive load. The results showed that the cognitive load does not differ significantly between all three groups. However, the differences in extraneous cognitive load were only just non-significant. The descriptive data analysis indicated that the control group had the highest extraneous and the lowest germane cognitive load. The group with the faded WE had the lowest extraneous cognitive load and a similar germane cognitive load to the non-faded WE group. Here too, the differences were only just above the significance level. The results on a descriptive level are consistent with the results of the existing literature. According to both Renkl (2014) and Paas et al. (2003), WE can help to reduce extraneous cognitive load and, in turn, lead to an increase in germane cognitive load.

With regard to the two hypotheses, the results of the data analysis only partially confirmed the first hypothesis. It was confirmed that the integration of faded as well as non-faded WE leads to a higher acquisition of applied knowledge compared to the control group without WE. An increase in factual knowledge was not found, however. Only at the descriptive level was a higher motivation and a different cognitive load measured. The second hypothesis could not be completely confirmed. No significant differences were found between the two WE conditions with regard to acquisition of factual and applied knowledge, motivation and lower extraneous cognitive load as well as higher germane cognitive load. The hypothesis was only confirmed with regard to the intrinsic cognitive load, as no differences were found here either. This could be due to the high complexity of the study topic (water body structure mapping). The topic is particularly difficult for beginners to handle, so learners could benefit from continuous support in the form of non-faded WE. A more detailed description of the results including statistical parameters can be found in the article in chapter 4.

In summary, the results of Study II suggest that WE are appropriate in supporting the application of scientific observation steps. Since no general advantage of faded WE over non-faded WE was shown, it is important to take into consideration both faded and non-faded WE. This information can be used to develop targeted interventions to support scientific observation competence.

8 General discussion and conclusion

8.1 Interpretation of empirical findings

The overall goal of this dissertation was to show how (mobile) digital media are used in informal learning contexts to support learning and to investigate cognitive and motivational effects. As there was no systematic overview of this relatively new field of research at the time of the study, in Study I the current state of research was analyzed by conducting a systematic literature review along the PRISMA guidelines in order to gain an initial overview of the use of (mobile) digital media to support informal learning with a focus on institutional informal learning places. The main aim here was to provide an overview of the general characteristics of (mobile) digital media in institutional informal learning places, their learning-relevant functions and the learning outcomes by using these media. In Study II, based on the findings of the systematic literature review, a mobile digital learning environment was developed and analyzed in an informal learning context to investigate via mixed method approach the effects of different forms of digitally presented WE (no WE vs. non-faded WE vs. faded WE) in the form of videos. In addition to the transfer of learning-relevant knowledge, the motivation and cognitive load of the students were also measured. Based on the results of Study II, the theoretical findings of Study I were expanded to include practical implementation in order to gain a holistic insight into learning with (mobile) digital media in institutional informal learning places using this combined perspective. The combined perspective of the studies of factual and applied knowledge as well as motivational and cognitive aspects contributes to this.

The starting point for the study was, on the one hand, the fact that learning in institutional informal learning places such as museums or zoos has recently become an increasing focus of the educational debate (Schwan et al., 2014). On the other hand, the increasing technological progress is also advancing in the field of education. Combining these two facts appeared to be useful, as digital media have already proven to be a suitable tool for supporting and promoting informal learning processes (Moser, 2017). This is particularly necessary because informal learning is often casual and unstructured (Lewalter & Neubauer, 2019), so support is required for learning success.

The results of Study I showed that the variety of informal learning places in the studies analyzed has increased over the years and that, in addition to traditional learning places such as museums, galleries or outdoor nature are increasingly being used as institutional informal learning places. Study II builds on this, in which outdoor nature is used as an informal learning framework in the context of water body structure mapping.

The analysis of the general characteristics of digital media in institutional informal learning places in Study I revealed that over time the design of digital media tools has become increasingly complex, such as 3D animations. In the last three years digital media tools have mainly been used in portable form, for example, in the form of smartphones or tablets. For Study II portable media tools in the form of tablets were used. Portable media tools can offer greater flexibility, especially in learning places such as outdoor nature where the use of stationary digital media such as tabletops is not possible or not appropriate in the learning setting. The evaluation of the functions of digital media tools in Study I showed that the provision of information and tasks are the main functions available, which corresponds to the results of Kampschulte et al. (2019). This shows that the full range of possible functions that digital media tools can provide is not yet fully used. A specific support function, for example, for adapting the task to the learner's level of knowledge was not found in the studies. Nevertheless, it is precisely this adaptability that is one of the strengths of digital media tools (Bannert & Reimann, 2009). Functions that explicitly support collaboration in groups were also very rarely found in the studies. However, the results of Study I showed that institutional informal learning places were mostly visited in groups. Studies on museum research indicate that museums are usually visited in social groups of families, friends or colleagues, so that the reception of the information is determined to a large extent from social interactions (Gutwill & Allen, 2010 in Schwan & Lewalter, 2019). For this reason, it seems useful to implement functions that support collaboration more strongly in the digital media in institutional informal learning places.

Overall, it can be stated that the number of functions that digital media tools can offer will increase steadily from 2018. With the increase and availability of technological progress, it could therefore be that the possible functions of digital media will continue to expand in the future and, as a result, can be further investigated. This gives rise to a need for further research. Despite the possible increase in functions that digital media can offer, the pedagogical perspective must always be considered. The use of various functions of a digital tool always depends on the (learning) context and the purpose for which the functions are used. This is, for example, the case with the digital learning environment developed for Study II, where the two main functions were providing information and tasks. Functions that promote social interaction were explicitly not intended, as this would bias the data.

One advantage of presenting information via digital media is the ability to easily provide additional and brief specific information, which can be adapted or even individualized as required (Schwan et al., 2018). Also, the information can be presented in different ways. This advantage was utilized when creating the digital learning environment for Study II in order to present the information in the form of videos in addition to the written instructions. In addition to the audio, images and graphics could be added for the visual component.

24

What Study I showed is that in museums the descriptions of exhibits are often placed next to the artefacts in the form of text displays. This can be suboptimal for the presentation of information, as it overloads the cognitive resources required to process the learning content (Schwan et al., 2018; Sweller et al., 2011). This problem represents the split-attention effect (Sweller et al., 2011), as the learner's visual attention is divided between viewing the picture and reading the text on the screen. One solution to this is the modality effect of the theory of multimedia learning (Mayer & Moreno, 2003). According to this, there is a better transfer of learning when the text is presented audibly as a narrative and not as written text on the screen. By using narrated text with pictures, part of the processing requirements can be transferred from the visual channel to the verbal channel. Schwan et al. (2018) came to the same result: when pictures are accompanied by audio instead of written text, learners develop a more elaborate mental representation of the picture content. This happens by focusing attention on relevant parts of the picture and then transferring them to working memory and by integrating the picture sor objects, for example, this could be realized using audio guides.

Based on the discussion of this result, it was shown how important it is to design the learning materials using the appropriate form of presentation and, in this context, to assess the cognitive load. However, the evaluation of the learning outcomes in Study I showed that this important aspect of cognitive load was not considered or documented in the studies analyzed. This is relevant insofar as the measurement of cognitive load has become a standard procedure in most studies concerned with digital media for learning objectives and is relevant for learning processes, as different design features of digital learning media can have both positive and negative effects on the cognitive processing of information (extraneous and germane load) (Skulmowski & Man Xu, 2021).

In order to ensure that the learning processes in the self-developed learning environment in Study II were not cognitively overloaded by the WE, the cognitive load was measured. Even though the results were just below the significance level, it was shown in consistency with the literature (Renkl, 2014; Paas et al., 2003) that the video-based WE can contribute to reducing the extraneous cognitive load and, in turn, lead to an increase in the germane cognitive load. The results once again highlight the importance of measuring cognitive load in order to be able to evaluate whether the digital tool has been designed adequate to enhance learning.

The further evaluation of the learning outcomes of Study I showed that the use of digital media in informal learning places have a positive effect on informal learning, in particular on knowledge acquisition, motivation, interest and engagement in the context of collaboration and social interaction. The positive results in terms of motivation can possibly be explained by the different types of presentation that digital media offer.

Through the use of AR, media can promote learning motivation and success and improve the flexibility and interactivity of learning activities (Lee et al., 2011). Such different presentation features can support learning on both a cognitive and motivational level (Bannert & Reimann, 2009).

Due to the positive findings from Study I with regard to knowledge, motivation and interest, these variables were also collected in Study II in order to investigate how WE, implemented in a digital learning environment, support the learning processes regarding to these variables. With regard to knowledge, the results showed that WE promote the acquisition of applied knowledge about scientific observation and that both versions of WE (non-faded and faded) are appropriate for this purpose. In contrast to the relevant literature (Renkl, 2014; Renkl et al., 2000), there was no difference between the three groups in the increase in factual knowledge about scientific observation. However, since WE have proven to be helpful in the acquisition of factual knowledge, especially in the version of faded WE, the significance of the study results with regard to factual knowledge about scientific observation should be examined here. As the students already achieved very good results in the pre-test on scientific observation, although half of the students indicated that they were novices in the field of scientific observation, it is possible that the level of difficulty of some test items may have been too low, leading to a biased presentation. On a descriptive level, it was shown that there was a higher level of motivation in the two groups that learned scientific observation with the WE. As the usefulness and overall learning with WE were assessed positively by the students, this could be an indication of the positive learning support provided by WE.

Overall, it has been shown that WE have a positive effect on the application of scientific observation about the scientific observation steps. By investigating applied knowledge using qualitative data analysis, it was possible to differentiate between the two types of knowledge - factual and applied knowledge - within the learning domain. A key implication of these results is that both faded and nonfaded WE need to be considered, as no general advantage of faded WE over non-faded WE was found. This information can be used to develop specific interventions that aim to enhance scientific observation competence. In conclusion, both studies show that the use of digital media in an informal learning context, in both institutional and formal settings, is appropriate for achieving positive results in terms of knowledge gain and motivation.

8.2 Limitations and suggestions for future research

The limitations of the two studies mentioned here provide indications for future research. What is the case with most systematic literature reviews is the limitation of the data material to be examined. In the systematic literature review (Study I), the search for appropriate studies was limited to the two databases Scopus and FIS, through which the three databases Eric, EBSCOhost eBooks and BASE were also analyzed.

The databases were chosen, because they offer the most literature references in the field of educational science. Scopus is an abstract and citation database for peer-reviewed literature, including scientific journals, books and conference papers. The database provides a comprehensive overview of global research output in the fields of science, technology, medicine, social sciences, arts and humanities (Scopus, o.D.). The FIS (Fachinformations-System Bildung) literature database offers literature references on all areas of education and currently contains over one million data records, such as monographs, anthologies and articles from specialist journals (Fachportal Pädagogik, o.D.). The search engine BASE (Bielefeld Academic Search Engine) collects references to electronically available research documents from disciplinary and institutional repositories worldwide. EBSCOhost eBooks contains ebooks from small, international publishers with currently around 400,000 predominantly English-language (book) titles based on the BISAC Subject Classification "Education" for all sub-disciplines of educational science, including higher education research. ERIC (Institute of Education Science) is the world's most comprehensive bibliographic database relevant to education and educational science with around 1.7 million journal articles, research and conference papers in English (Fachportal Pädagogik, o.D.)

In addition to the limitation of the search location, the time at which the search was carried out also represents a limitation. The literature search was conducted between December 2020 and February 2021, which means that more recent studies and thus other potentially important findings were not included in the evaluation process. Despite the limited number of articles included in the literature review it must be said that the selection was made according to a systematic procedure in order to avoid bias. As the literature review was conducted three years ago, it would be interesting for future research to conduct another systematic literature review, including other databases. This would provide an overview of what has changed over the past three years with regard to the advancing digitalization of institutional informal learning places. One point to consider when interpreting the results is the consistently positive findings of the study. Only one study (Dieck et al., 2016) mentions disadvantages with regard to the use of digital media in institutional informal learning places. A publication bias cannot be ruled out here, as positive results are easier to publish and these are peerreviewed articles (Sharma & Verma, 2019).

When designing the search syntax, only some quite well-known institutional informal learning places such as museums, aquariums and zoos were included in the search syntax. However, the general search keywords led to articles involving less well-known institutional informal learning places such as science centers, botanical gardens or nature parks. As the results of the systematic review show that there are many other institutional informal learning places beyond the common learning places, it would be useful for future research to consider this diversity of institutional informal learning places more closely.

27

The sample size in the Study II represents a further limitation in terms of the generalizability of the results. Nevertheless, a small sample size was necessary in order to be able to conduct the comprehensive qualitative data analysis that was important for this study. With regard to the descriptive tendency, a larger sample might yield significant results in future research and reveal even small effects. In this study, this is the case for cognitive load, where the differences were just above the significance level. The sample also consisted only of students, particularly from the natural sciences. In a future investigation, the study could be repeated with students from other disciplines to increase generalizability. A further limitation of the Study II is the fact that the participants already performed very high in the pre-test on scientific observation, so that participation in the intervention would probably not result in a significant increase in knowledge due to ceiling effects (Staus et al., 2021). Nevertheless, almost half of the students reported that they were novices in the field before the study, which indicates that the difficulty level of some test items may have been too low. Future research could revise the factual knowledge test, for example, the difficulty of the distractors, and test it again in order to obtain more valid results regarding the students' knowledge gain about scientific observation. Another limitation is that only one exemplary topic was examined in Study II. Only the learning domain of the dual content WE were investigated, so that in future research the exemplifying domain could be analyzed or other variables such as learning and performance-related attitudes could be included. Future research could investigate the influence of learners' prior knowledge on learning with WE as studies have shown that WE are particularly beneficial in the initial acquisition of cognitive skills (Kalyuga et al., 2001).

A final suggestion for future research concerns both studies. In both studies analyzed in the literature review and in the empirical study, knowledge and interest were only tested immediately after completion of the treatment and no follow-up test was reported or conducted. It would be interesting to conduct further research into the long-term effects of knowledge and interest gain when learning with digital media.

8.3 Implications for research and practice

The present dissertation extends previous research on informal learning with Study I by providing, for the first time, a systematic overview of learning with digital media in institutional informal learning places and showing how (mobile) digital media are used to support informal learning with a focus on institutional informal learning places. In addition to an initial overview of the characteristics of these learning places, the functions of digital media and their learning outcomes are examined in particular. In this way, Study I provides important findings for media-based educational research. The results of Study I have several practical implications for the use of digital media in institutional informal learning places. In particular, the design of digital media in terms of their functions as well as motivational and cognitive aspects should be emphasized. The question of what added value digital media can offer for learning in these places of learning needs to be asked. In addition to classic functions such as the provision of information and tasks, however, it is primarily the functions that can support the learning processes that were found very rarely or not at all in the studies. None of the studies analyzed offered the possibility of adapting the learning tasks to the learner's level of knowledge (Gerard et al., 2015), although this type of support function is one of the greatest strengths of digital media tools (Bannert & Reimann, 2009). There is a need for further studies to include this function more in the design of digital media tools. The same applies to the collaboration functions, as the qualitative analysis show that institutional informal learning places were often visited in groups. For example, by playing learning games or quizzes together knowledge could be gained in a playful way through collaborative exchange.

Further action is required with regard to the measurement of cognitive load. The results of Study I revealed that cognitive load was not recorded or reported in the studies analyzed, meaning that no assumptions can be made about the extent to which this could have had a negative impact on the cognitive processing of information relevant to learning. Furthermore, the explicit design of the digital media tools could not be investigated, as the design characteristics were often not mentioned in the analyzed studies. For further research, it could be interesting to analyze the design characteristics of the media tools in greater detail by looking into the design characteristics of the cognitive theory of multimedia (Mayer & Moreno, 2003) (for example, modality effect, chapter 8.1) in order to examine aspects such as the cognitive load, especially the external and internal load (Skulmowski & Man Xu, 2021), or the format of the instructions. This may be especially important in the context of the specific support function through adaptivity and interactivity to provide a tool for active learning (Bannert & Reimann, 2009; Gerard et al., 2015). Interactivity as a support function with a low or medium level of interactivity can enhance learning and motivation while avoiding cognitive overload (Skulmowski & Man Xu, 2021). Interactivity is hardly possible without a minimum level of adaptivity. That means, that statements made by the learning systems should relate to previous statements made by the user and be adapted to the specific characteristics (for example, prior knowledge, interests, learning speed, language) of the individual learner as far as possible (Niegemann & Heidig, 2019). Adaptive guidance is more effective than guidance in traditional learning contexts, especially for learners with little prior knowledge. Adaptivity is most effective when learners generate and integrate ideas, for example, when creating concept diagrams, rather than just selecting from given options. Guidance that encourages self-monitoring is more likely to improve learning outcomes than guidance that focuses only on content knowledge (Gerard et al., 2015).

Study II provides important findings for media-based educational research and for learning in an informal learning context, but in a formal setting. From a theoretical perspective, the results partly replicate the findings of existing studies and represent an extension of previous research. On a descriptive level it has been shown that even in an informal learning context but formal setting, WE lead to higher learning-relevant motivation (Paas et al., 2005; Van Harsel et al., 2019), reduces extrinsic cognitive load and increases germane cognitive load (Paas et al., 2003; Renkl, 2014) and supports learning in the faded version (Hesser and Gregory, 2015; Renkl, 2014). It has been shown that WE significantly enhance the acquisition of applied knowledge. Contrary to what was to be expected from the literature, an increase in learning in acquisition of factual knowledge was not found and faded WE did not prove to be more effective than non-faded WE for the acquisition and transfer of factual knowledge (Renkl, 2014; Renkl et al., 2000). One explanation for this could be the ceiling effects described in chapter 8.2. In addition, the results confirm the existing evidence that WE can be useful for building scientific competences. Further research could focus on investigating the exemplifying domain in more detail to test whether non-faded WE are more effective for learning than faded WE in the case of WE with double-content and complex topics such as scientific observation and water body structure mapping.

8.4 Conclusion

The aim of this dissertation was to investigate how (mobile) digital media are used in informal learning contexts and to what extent the media can support the informal learning in terms of cognitive and motivational aspects. For the first time ever, a systematic literature review of this young field of research was conducted. Study I showed that learning with digital media in institutional informal learning places is appropriate for fostering knowledge acquisition, motivation, interest and engagement. Due to the variety of functions that digital media can offer, they have great potential to structure and support learning, even if they do not yet fully exploit this potential due to the fact that they are often limited to the provision of information and tasks. Functions involving interactivity and adaptivity in particular could be given more attention in the future design of digital media in order to adapt learning even more to the individual needs of learners. There is a need for further research with regard to the measurement of cognitive load, as no information was given in the studies analyzed. As cognitive load can have positive as well as negative impact on the cognitive processing of information, particularly in the case of extraneous and germane load, further research in this field would help to design learning with digital media in institutional informal learning places in such a way that it is not cognitively overloaded.

The results of Study II provide important findings for the application of double-content WE to develop scientific observation competence. Even though only the learning domain was investigated, the analysis of the combined perspective of factual and applied knowledge represents a further added value of the study. In contrast to the existing literature, no general advantage of faded WE over non-faded WE was found, which suggests that WE, not only in their faded version but also in their non-faded version are appropriate to support learners' applied knowledge when learning about scientific observation. These results can be used to develop specific interventions aimed to enhance and promote scientific observation competence. Further research could help to investigate the motivation for learning with double-content WE more precisely as the study results could only confirm the hypotheses on a descriptive level.

The dissertation gives important findings for informal learning and for media-based educational research. The dissertation expands existing research by providing for the first time a comprehensive, and systematic literature overview on the characteristics, functions and learning outcomes of and with digital media in institutional informal learning places. The combined perspective of factual and applied knowledge in the learning domain of double-content WE also provide an initial starting point for this research field to consider and analyze the acquisition of different types of knowledge during learning with double-content WE more closely. Overall, the results can contribute by showing how digital media should be designed in self-directed educational contexts to promote the acquisition of factual and applied knowledge and to be as motivating and cognitively less loaded as possible.

References

- Ammad-ud-din, M., Mikkonen, T., Pinjamaa, N., Lehto, S., Ståhlberg, P., & Ventura, E. (2014). How will digital media impact education? In Y. Neuvo, E. Ormala, & M. Kuikka (Eds.), *Bit Bang 6: Future* of Media (pp. 141-163). Aalto ARTS Books.
- Aspers, P., & Corte, U. (2019). What is Qualitative in Qualitative Research. *Qualitative Sociology, 42*, 139 160. <u>https://doi.org/10.1007/s11133-019-9413-7</u>
- Atkinson, R. K., Derry, S. J., Renkl, A., & Wortham, D. (2000). Learning from Examples: Instructional Principles from the Worked Examples Research. *Review of Educational Research*, *70*(2), 181–214. <u>https://doi.org/10.3102/00346543070002181</u>
- Bannert, M., & Reimann, P. (2009). *Effects of orientation prompts when learning with Hypermedia*. Paper presented at the AERA 09-Annual Meeting Conference, San Diego, USA.
- Barbieri, C.A., Booth, J.L., Begolli, K.N., & McCann, N. (2021). The effect of worked examples on student learning and error anticipation in algebra. *Instructional Science*, 49(4), 419-439. <u>https://doi.org/10.1007/s11251-021-09545-6</u>
- Bayerisches Landesamt für Umwelt. (2019). *Gewässerstrukturkartierung von Fließgewässern in Bayern* – Erläuterungen zur Erfassung und Bewertung. (Water structure mapping of flowing waters in Bavaria - Explanations for recording and assessment) <u>https://www.bestellen.bayern.de/application/eshop_app000005?SID=1020555825&ACTION</u> <u>xSESSxSHOWPIC(BILDxKEY:%27lfu_was_00152%27,BILDxCLASS:%27Artikel%27,BILDxTYPE:%2</u> <u>7PDF%27)</u>
- Berthold, K., Eysink, T. H., & Renkl, A. (2009). Assisting self-explanation prompts are more effective than open prompts when learning with multiple representations. *Instructional Science*, *37*, 345–363. <u>https://doi.org/10.1007/s11251-008-9051-z</u>
- *Block, F., Hammerman, J., Horn, M., Spiegel, A., Christiansen, J., Phillips, B., Diamond, J., Evans, M. E., & Shen, C. (2015). *Fluid Grouping: Quantifying Group Engagement around Interactive Tabletop* <u>Exhibits in the Wild.</u> In Proceedings of the 2015 ACM annual conference on Human Factors in Computing Systems.
- Bokosmaty, S., Sweller, J., & Kalyuga, S. (2015). Learning geometry problem solving by studying worked examples: Effects of learner guidance and expertise. *American Educational Research Journal*, 52(2), 307–333. <u>https://doi.org/10.3102/0002831214549450</u>
- Booth, J.L., McGinn, K., Young, L.K., & Barbieri, C.A. (2015). Simple Practice Doesn't Always Make Perfect. *Policy Insights from the Behavioral and Brain Sciences, 2*(1), 24 - 32. https://doi.org/10.1177/2372732215601691
- Brownell, S. E., Wenderoth, M. P., Theobald, R., Okoroafor, N., Koval, M., Freeman, S., Walcher-Chevillet, C. L., & Crowe, A. J. (2013). How Students Think about Experimental Design: Novel Conceptions Revealed by in-Class Activities. *BioScience*, 64(2), 125–137. https://doi.org/10.1093/biosci/bit016
- Chien, J. (2012). *How digital media and Internet is transforming education*. In Proceedings of the 2012 AACE, E-Learning conference.
- Chinn, C. A., & Malhotra, B. A. (2001). Epistemologically authentic scientific reasoning. In K. Crowley, C. D. Schunn, and T. Okada (Eds.). *Designing for science: Implications from everyday, classroom, and professional settings* (pp. 351–392). Lawrence Erlbaum.

- *Choi, W. G. Land, S. M., & Zimmerman, H. T. (2018). Investigating children's deep learning of the tree life cycle using mobile technologies. *Computers in Human Behavior*, 87, 470-479. <u>https://doi.org/10.1016/j.chb.2018.04.020</u>
- *Civantos, A. M., Brown, M., Coughlan, T., Ainsworth, S., & Lorenz, K. (2016). Using mobile media creation to structure museum interpretation with professional vision. *Personal and Ubiquitous Computing*, 20(1), 23–36. <u>https://doi.org/10.1007/s00779-015-0895-3</u>
- *Clayphan, A., Collins, A., Kay, J., Slawitschka, N., & Horder, J. (2018). Comparing a Single-Touch Whiteboard and a Multi-Touch Tabletop for Collaboration in School Museum Visits. Proceedings of the ACM on Interactive, Mobile, Wearable and Ubiquitous Technologies, 2(1), 1–23. <u>https://doi.org/10.1145/3191738</u>
- Dart, S., Pickering, E., & Dawes, L. (2020). Worked Example Videos for Blended Learning in Undergraduate Engineering. AEE Journal, 8(2), 1-22. <u>http://dx.doi.org/10.18260/3-1-1153-36021</u>
- Dasgupta, A. P., Anderson, T. R., & Pelaez, N. (2014). Development and Validation of a Rubric for Diagnosing Students' Experimental Design Knowledge and Difficulties. CBE—Life Sciences Education, 13(2), 265–284. <u>https://doi.org/10.1187/cbe.13-09-0192</u>
- Deane, T., Nomme, K., Jeffery, E., Pollock, C., & Birol, G. (2014). Development of the Biological Experimental Design Concept Inventory (BEDCI). *CBE—Life Sciences Education*, *13*(3), 540–551. https://doi.org/10.1187/cbe.13-11-0218
- Deci, E. L., & Ryan, R. M. (2012). Self-determination theory. In P. A. M. Van Lange, A. W. Kruglanski, and E. T. Higgins (Eds.), *Handbook of theories of social psychology* (pp. 416–436). SAGE.
- Deutscher Bildungsrat. (1974): Zur Neuordnung der Sekundarstufe II: Konzept für eine Verbindung von allgemeinem und beruflichem Lernen. Empfehlungen der Bildungskommission. Klett.
- *Dieck, M.C., Jung, T., & Dieck, D.T. (2016). Enhancing art gallery visitors' learning experience using wearable augmented reality: generic learning outcomes perspective. *Current Issues in Tourism*, 21(17), 2014–2034. <u>https://doi.org/10.1080/13683500.2016.1224818</u>
- *Eberbach, C., & Crowley, K. (2005). From Living to Virtual: Learning from Museum Objects. *The Museum Journal, 48*(3), 317–338. <u>https://doi.org/10.1111/j.2151-6952.2005.tb00175.x</u>
- Eberbach, C. & Crowley, K. (2009). From Everyday to Scientific Observation: How Children Learn to Observe the Biologist's World. *Review of Educational Research*, 79(1), 39-68. <u>https://doi.org/10.3102/0034654308325899</u>
- Eshach, H. (2007). Bridging in-school and out-of-school learning: Formal, non-formal, and informal education. *Journal of Science Education and Technology*, 16(2), 171–190. https://doi.org/10.1007/s10956-006-9027-1

European Commission. (2002). A European Area of Lifelong Learning. European Commission.

- *Evans, M.A., Lopez, M., Maddox, D., Drape, T., & Duke, R.F. (2014). Interest-Driven Learning Among Middle School Youth in an Out-of-School STEM Studio. *Journal of Science Education and Technology, 23*(5), 624–640. <u>https://doi.org/10.1007/s10956-014-9490-z</u>
- FachportalPädagogik.(o.D.).FISBildung.https://www.fachportal-paedagogik.de/literatur/produkte/fis_bildung/fis_bildung.html
- Falk, J. H. (2009). Identity and the museum visitor experience. Left.
- Ford, D. (2005). The challenges of observing geologically: Third graders' descriptions of rock and mineral properties. *Science Education*, *89*(2), 276–295. <u>https://doi.org/10.1002/sce.20049</u>

- *Fränkel, S., Sellmann-Risse, D., & Grotjohann, N. (2020). Digitale Schnitzeljagd im Wald Der Actionbound "Dem Wald auf der Spur" zur Förderung digitaler und fachlich-biologischer Kompetenzen. *Herausforderung Lehrer_innenbildung, 3*(1), 34–48. https://doi.org/10.4119/hlz-2526
- Gerard, L., Matuk, C., McElhaney, K., & Linn, M. C. (2015). Automated, adaptive guidance for K-12 education. *Educational Research Review*, 15, 41-58. https://doi.org/10.1016/j.edurev.2015.04.001
- Gerber, B. L., Cavallo, A. M. L., & Marek, E. M. (2001). Relationships among informal learning environments, teaching procedures and scientific reasoning ability. *International Journal of Science Education*, 23(5), 535–549. <u>https://doi.org/10.1080/09500690116971</u>
- Gerjets, P., Scheiter, K., & Catrambone, R. (2004). Designing Instructional Examples to Reduce Intrinsic Cognitive Load: Molar versus Modular Presentation of Solution Procedures. *Instructional Science*, 32(1/2), 33–58. <u>https://doi.org/10.1023/b:truc.0000021809.10236.71</u>
- Gupta, U. (2019). Interplay of Germane Load and Motivation During Math Problem Solving using Worked Examples. *Educational Research: Theory and Practice*, *30*(1), 67-71.
- Gutwill, J. P., & Allen, S. (2010). Facilitating family group inquiry at science museum exhibits. *Science Education*, 94(4), 710–742. <u>https://doi.org/10.1002/sce.20387</u>
- *Hammady, R., Ma, M., Strathern, C., & Mohamad, M. (2019). Design and development of a spatial mixed reality touring guide to the Egyptian museum. *Multimedia Tools and Applications*, 79(5-6), 3465-3494. <u>https://doi.org/10.1007/s11042-019-08026-w</u>
- Hawkey, R. (2004). *Learning with Digital Technologies in Museums, Science Centres and Galleries*. A NESTA Futurelab Series Report 7.
- Hefter, M. H., Berthold, K., Renkl, A., Riess, W., Schmid, S., & Fries, S. (2014). Effects of a training intervention to foster argumentation skills while processing conflicting scientific positions. *Instructional Science*, 42(6), 929–947. <u>https://doi/10.1007/s11251-014-9320-y</u>
- Hesser, T.L., & Gregory, J.L. (2015). Exploring the Use of Faded Worked Examples as a Problem Solving Approach for Underprepared Students. *Higher Education Studies*, 5(6), 36-46. <u>https://doi.org/10.5539/hes.v5n6p36</u>
- Higgins, J. P. T., Thomas, J., Chandler, J., Cumpston. M., Li, T., Page, M. J., & Welch, V. A. (Eds.). (2019). *Cochrane handbook for systematic reviews of interventions* (2nd ed.). Wiley. <u>https://doi.org/10.1002/9781119536604</u>
- *Horn, M., Phillips, B., Evans, E., Block, F., Diamond, J., & Shen, C. (2016). Visualizing biological data in museums: Visitor learning with an interactive tree of life exhibit. *Journal of Research in Science Teaching*, 53(6), 895-918. <u>https://doi.org/10.1002/tea.21318</u>
- Ibáñez, M., & Delgado-Kloos, C. (2018). Augmented reality for STEM learning: A systematic review. *Computers in Education, 123*, 109-123. <u>https://doi.org/10.1016/j.compedu.2018.05.002</u>
- Jensen, E. (2014). Evaluating children's conservation biology learning at the zoo. *Conservation Biology*, 28(4), 1004-1011. <u>https://doi.org/10.1111/cobi.12263</u>
- Kalyuga, S., Ayres, P., Chandler, P., & Sweller, J. (2003). The Expertise Reversal Effect. *Educational Psychologist*, *38*(1), 23 31. <u>https://doi.org/10.1207/s15326985ep3801_4</u>
- Kalyuga, S., Chandler, P., Tuovinen, J. & Sweller, J. (2001). When problem solving is superior to studying worked examples. *Journal of Educational Psychology*, 93(3), 579–588. <u>https://doi/10.1037/0022-0663.93.3.579</u>

- Kampschulte, L., Ostermann, A., Müller, F., Ropohl, M., Schwanewedel, J., Härtig, H. & Lindmeier, A. (2019). Einsatz digitaler und analoger Medien an außerschulischen Lernorten. IPN – Leibniz-Institut für die Pädagogik der Naturwissenschaften und Mathematik.
- Kay, R.H., & Edwards, J. (2012). Examining the Use of Worked Example Video Podcasts in Middle School Mathematics Classrooms: A Formative Analysis. *Canadian Journal of Learning and Technology*, 38(3). <u>https://doi.org/10.21432/t2pk5z</u>
- Kerres, M. (2000). Mediendidaktische Analyse digitaler Medien im Unterricht. *Computer + Unterricht :* Lernen und Lehren mit digitalen Medien, 10(1), 26 – 28.
- *Kersting, M., Steier, R., & Venville, G. (2020). Exploring participant engagement during an astrophysics virtual reality experience at a science festival. *International Journal of Science Education, Part B*, 11(1), 17–34. <u>https://doi.org/10.1080/21548455.2020.1857458</u>
- Khaddage, F., Müller,W., &Flintoff, K.(2016). Advancing Mobile Learning in Formal And Informal Settings via Mobile App Technology: Where to From Here, and How? *Educational Technology & Society*, 19(3), 16–26.
- Klepsch, M., Schmitz, F. & Seufert, T. (2017). Development and Validation of Two Instruments Measuring Intrinsic, Extraneous, and Germane Cognitive Load. Frontiers in Psychology, 8. <u>https://doi.org/10.3389/fpsyg.2017.01997</u>
- Knoblich, L. (2020). Digital gestützte Biodiversitätsexkursionen. *Biologie Unserer Zeit, 50*(2), 134-142. https://doi.org/10.1002/biuz.202010703
- Knogler, M., Harackiewicz J. M., Gegenfurtner, A., & Lewalter, D. (2015). How situational is situational interest? Investigating the longitudinal structure of situational interest. *Contemporary Educational Psychology*, 43, 39-50. <u>https://doi/10.1016/j.cedpsych.2015.08.004</u>
- Koenen, J. (2014). Entwicklung und Evaluation von experimentunterstützten Lösungsbeispielen zur Förderung naturwissenschaftlich experimenteller Arbeitsweisen. Dissertation. Universität Duisburg-Essen.
- Koenen, J., Emden, M. & Sumfleth, E. (2017). Naturwissenschaftlich-experimentelles Arbeiten. Potenziale des Lernens mit Lösungsbeispielen und Experimentierboxen. (Scientificexperimental work. Potentials of learning with solution examples and experimentation boxes). Zeitschrift für Didaktik der Naturwissenschaften, 23(1), S. 81-98. https://doi.org/10.1007/s40573-017-0056-5
- Kohlhauf, L., Rutke, U., & Neuhaus, B. J. (2011). Influence of Previous Knowledge, Language Skills and Domain-specific Interest on Observation Competency. *Journal of Science Education and Technology*, 20(5), 667. <u>https://doi.org/10.1007/s10956-011-9322-3</u>
- Lee, H., Billinghurst, M., & Woo, W. (2011). Two-handed tangible interaction techniques for composing augmented blocks. *Virtual Reality*, *15*(2-3), 133-146. <u>https://doi.org/10.1007/s10055-010-0163-9</u>
- Leibniz-bildungspotenziale@dipf.de. (2023, October 16). Positionspapier des Leibniz-Forschungsnetzwerks Bildungspotenziale (LERN) zum Thema "Außerschulische und informelle Lernorte für Kinder und Jugendliche". <u>https://www.leibniz-bildung.de/wpcontent/uploads/2021/07/BPF_Positionspapier_Gesamt_FINAL_20230920.pdf</u>
- Leppink, J., Paas, F., Van der Vleuten, C. P., Van Gog, T., & Van Merriënboer, J. J. (2013). Development of an instrument for measuring different types of cognitive load. *Behavior research methods*, 45(4), 1058-1072. <u>https://doi.org/10.3758/s13428-013-0334-1</u>

- Lewalter, D. (2020). Schülerlaborbesuche aus motivationaler Sicht unter besonderer Berücksichtigung des Interesses. (Student laboratory visits from a motivational perspective with special attention to interest). In: Sommer, K., Wirth, J. and Vanderbeke, M. (Hrsg.) Handbuch Forschen im Schülerlabor Theoretische Grundlagen, empirische Forschungsmethoden und aktuelle Anwendungsgebiete (S. 62-70). Waxmann-Verlag.
- Lewalter, D. & Knogler, M. (2014, April). A questionnaire to assess situational interest theoretical considerations and findings. Poster presented at the 50th Annual Meeting of the American Educational Research Association (AERA), Philadelphia, USA.
- Lewalter, D., & Neubauer, K. (2020). Stichwort: Informelles Lernen im Kindes- und Jugendalter. *Psychologie in Erziehung und Unterricht, 67*(3), 228-231. <u>https://doi.org/10.2378/peu2020.art18d</u>
- Lewalter, D., & Neubauer, K. (2019). Informelles Lernen. In D. Urhhane, M. Dresel & F. Fischer (Eds.), *Psychologie für den Lehrberuf* (pp. 126-142). Springer.
- Lewalter, D., & Willems, A.S. (2004). Die Bedeutung des motivationsrelevanten Erlebens und des individuellen Fachinteresses für das situationale Interesse im Mathematikunterricht. *Psychologie in Erziehung Und Unterricht, 56*(4), 243-257. Reinhardt Verlag.
- Lin, M.-H., Chen, H.-C., & Liu, K.-S. (2017). A Study of the Effects of Digital Learning on Learning Motivation and Learning Outcome. *Eurasia Journal of Mathematics, Science and Technology Education*, 13(7), 3553–3564. <u>https://doi.org/10.12973/eurasia.2017.00744a</u>
- Lin, P.-Y., & Schunn, C. D. (2016). The dimensions and impact of informal science learning experiences on middle schoolers' attitudes and abilities in science. *International Journal of Science Education*, 38(17), 2551–2572. <u>https://doi.org/10.1080/09500693.2016.1251631</u>
- Lunetta, V., Hofstein, A. & Clough, M. P. (2007). Learning and teaching in the school science laboratory: An analysis of research, theory, and practice. In N. Lederman, and S. Abel (Eds.). Handbook of research on science education (pp. 393-441). Lawrence Erlbaum.
- Mantiri, F. (2014). Multimedia and Technology in Learning. *Universal Journal of Educational Research*, 2(9), 589-592. <u>https://doi.org/10.13189/ujer.2020.081278</u>
- Mayer, R. E. (2001). *Multimedia learning.* Cambridge University Press. https://doi.org/10.1017/CBO9781139164603
- Mayer, R. E., & Moreno, R. (2003). Nine ways to reduce cognitive load in multimedia learning. *Educational Psychologist, 38*(1), 43–52. <u>https://doi.org/10.1207/s15326985ep3801_6</u>
- Mayring, P. (2015). Qualitative Inhaltsanalyse. Grundlagen und Techniken (12. Aufl.). Beltz Verlag.
- Mohajan, H. K. (2018). Qualitative Research Methodology in Social Sciences and Related Subjects. Journal of Economic Development, Environment and People, 7(1), 23-48. https://doi.org/10.26458/jedep.v7i1.571
- *Moorhouse, N., tom Dieck, M. C., & Jung, T. (2019). An experiential view to children learning in museums with Augmented Reality. *Museum Management and Curatorship*, 34(4), 402–418. <u>https://doi.org/10.1080/09647775.2019.1578991</u>
- Moser, S. (2017). Linking Virtual and Real-life Environments: Scrutinizing Ubiquitous Learning Scenarios. In S. Şad & M. Ebner (Eds.), *Digital Tools for Seamless Learning* (pp. 214-239). IGI Global. <u>https://doi.org/10.4018/978-1-5225-1692-7.ch011</u>
- Moser, S., Zumbach, J. & Deibl, I. (2017). The effect of metacognitive training and prompting on learning success in simulation-based physics learning. *Science Education*, 101(6), 944–967. https://doi.org/10.1002/sce.21295

- Morris, B. J., Owens, W., Ellenbogen, K., Erduran, S., & Dunlosky, J. (2019). Measuring informal STEM learning supports across contexts and time. *International Journal of STEM Education, 6*(1), 1-12. <u>https://doi.org/10.1186/s40594-019-0195-y</u>
- *Nelson, B. C., Bowman, C. D. D., Bowman, J. D., Pérez Cortés, L. E., Adkins, A., Escalante, E., Owen, B. L., Ha, J., & Su, M. (2019). Ask Dr. Discovery: the impact of a casual mobile game on visitor engagement with science museum content. *Educational Technology Research and Development*, 68(1), 345–362. <u>https://doi.org/10.1007/s11423-019-09696-x</u>
- Neubauer, K. (2015). Unterstützung naturwissenschaftlicher Grundbildung durch Schulklassenbesuche in naturwissenschaftlich-technischen Museen. Motivationale und kognitive Wirkung unterschiedlicher Besuchsformen (Dissertation). Technische Universität München, München.
- Niegemann, H., & Heidig, S. (2019). Interaktivität und Adaptivität in multimedialen Lernumgebungen. In H. Niegemann & A. Weinberger (Hrsg.), Handbuch Bildungstechnologie. Konzeption und Einsatz digitaler Lernumgebungen (S. 343-367). Springer. <u>https://doi.org/10.1007/978-3-662-54368-9_33</u>
- Paas, F., Renkl, A., & Sweller, J. (2003). Cognitive load theory and instructional design: Recent developments. *Educational psychologist*, *38*(1), 1–4. <u>https://doi.org/10.1207/S15326985EP3801 1</u>
- Paas, F., Tuovinen, J. E., van Merriënboer, J. J. G., & Aubteen Darabi, A. (2005). A motivational perspective on the relation between mental effort and performance: Optimizing learner involvement in instruction. *Educational Technology Research and Development*, 53(3), 25–34. <u>https://doi.org/10.1007/bf02504795</u>
- Petko, D. (2014). Einführung in die Mediendidaktik. Lehren und Lernen mit digitalen Medien. Beltz.
- Reiss, K. M., Heinze, A., Renkl, A., & Groß, C. (2008). Reasoning and proof in geometry: effects of a learning environment based on heuristic worked-out examples. *ZDM – The International Journal on Mathematics Education*, 40(3), 455–467. <u>http://dx.doi.org/10.1007/s11858-008-0105-0</u>
- Renkl, A. (2001). Explorative Analysen zur effektiven Nutzung von instruktionalen Erklärungen beim Lernen aus Lösungsbeispielen. *Unterrichtswissenschaft 29*(1), 41-63. <u>https://doi.org/10.25656/01:7677</u>
- Renkl, A. (2014). The worked examples principle in multimedia learning. In R. E. Mayer (Ed.), *Cambridge handbook of multimedia learning* (pp. 391–412). Cambridge University Press. <u>https://doi.org/10.1017/CB09781139547369.020</u>
- Renkl, A. (2017). Learning from worked-examples in mathematics: students relate procedures to principles. *ZDM The International Journal on Mathematics Education, 49*(4), 571-584. https://doi.org/10.1007/s11858-017-0859-3
- Renkl, A., Atkinson, R. K., & Große, C. S. (2004). How fading worked solution steps works. A Cognitive Load perspective. *Instructional Science, 32*(1-2), 59–82. <u>https://doi.org/10.1023/B:TRUC.0000021815.74806.f6</u>
- Renkl, A., Atkinson, R. K, & Maier, U. H. (2000). From Studying Examples to Solving Problem: Fading Worked-Out Solution Steps Helps Learning. *Proceedings of the Annual Meeting of the Cognitive Science Society*, 22. <u>https://escholarship.org/uc/item/81b9j9hs</u>
- Renkl, A., Atkinson, R. K., Maier, U. H., & Staley, R. (2002). From Example Study to Problem Solving: Smooth Transitions Help Learning. *The Journal of Experimental Education*, *70*(4), 293–315. <u>https://doi.org/10.1080/00220970209599510</u>

- Renkl, A., Hilbert, T., & Schworm, S. (2009). Example-based learning in heuristic domains: a cognitive load theory account. *Educational Psychology Review*, 21(1), 67–78. https://psycnet.apa.org/doi/10.1007/s10648-008-9093-4
- Rogers, A. (2014). *The Base of the Iceberg: Informal Learning and Its Impact on Formal and Non-formal Learning.* Verlag Barbara Budrich. <u>https://doi.org/10.3224/84740632</u>
- Schugurensky, D. (2000). *The forms of informal learning: Towards a conceptualization of the field* (WALL Working Paper No. 19). <u>https://tspace.library.utoronto.ca/bitstream/1807/2733/2/19formsofinformal.pdf</u>
- Schwan, S. (2015). Informelles Lernen in Museen und Science Centern. In M. Rohs (Hrsg.), Handbuch Informelles Lernen (S.379-396). Springer Verlag. https://doi.org/10.1007/978-3-658-06174-6_43-1
- *Schwan, S., Dutz, S., & Dreger, F. (2018). Multimedia in the wild: Testing the validity of multimedia learning principles in an art exhibition. *Learning and Instruction, 55*, 148-157. https://doi.org/10.1016/j.learninstruc.2017.10.004
- Schwan, S., Grajal, A., & Lewalter, D. (2014). Understanding and Engagement in Places of Science Experience: Science Museums, Science Centers, Zoos, and Aquariums. *Educational Psychologist*, 49(2), 70 - 85. <u>https://doi.org/10.1080/00461520.2014.917588</u>
- Schwan, S., Lewalter, D. (2020). Multimediales Lernen in öffentlichen Bildungseinrichtungen am Beispiel von Museen und Ausstellungen. In H. Niegemann, H. & A. Weinberger (Hrsg.), Handbuch Bildungstechnologie (S. 689-697). Springer Verlag. <u>https://doi.org/10.1007/978-3-662-54368-9_57</u>
- Schworm, S., & Renkl, A. (2007). Learning argumentation skills through the use of prompts for selfexplaining examples. *Journal of Educational Psychology*, *99*(2), 285–296. https://doi.org/10.1037/0022-0663.99.2.285
- Scopus. (o.D.). Scopus: Comprehensive, multidisciplinary, trusted abstract and citation database. https://www.elsevier.com/products/scopus?dgcid=RN_AGCM_Sourced_300005030
- Sefton-Green, J. (2004). *Literature review in informal learning with technology outside school.* A NESTA Futurelab Series Report 7. Bristol, UK: Futurelab. <u>https://telearn.archives-ouvertes.fr/hal-00190222/document</u>
- Sharma, H., & Verma, S. (2019). Is positive publication bias really a bias, or an intentionally created discrimination toward negative results? *Saudi Journal of Anaesthesia*, *13*(4), 352 355. <u>https://doi.org/10.4103/sja.sja_124_19</u>
- Siddaway, A. P., Wood, A. M., & Hedges, L. V. (2019). How to Do a Systematic Review: A Best Practice Guide for Conducting and Reporting Narrative Reviews, Meta-Analyses, and Meta-Syntheses. *Annual Review of Psychology*, 70(1), 747–770. https://doi.org/10.1146/annurev-psych-010418-102803
- Sirum, K., & Humburg, J. (2011). The Experimental Design Ability Test (EDAT). *Bioscene: The Journal Of College Biology Teaching, 8*(37), 8-16.
- Skulmowski, A., & Xu, K. M. (2021). Understanding Cognitive Load in Digital and Online Learning: a New Perspective on Extraneous Cognitive Load. *Educational Psychology Review*, *34*(1), 171–196. https://doi.org/10.1007/s10648-021-09624-7
- *Sommerauer, P., & Müller, O. (2014). Augmented reality in informal learning environments: A field experiment in a mathematics exhibition. *Computers & Eduction, 79,* 59-68. <u>https://doi.org/10.1016/j.compedu.2014.07.013</u>

- Staus, N. L., O'Connell, K., & Storksdieck, M. (2021). Addressing the Ceiling Effect when Assessing STEMOut-Of-SchoolTimeExperiences.FrontiersinEducation,6.https://doi.org/10.3389/feduc.2021.690431
- Sweller, J. (2006). The worked example effect and human cognition. *Learning and Instruction*, *16*(2) 165–169. <u>https://psycnet.apa.org/doi/10.1016/j.learninstruc.2006.02.005</u>
- Sweller, J. (2010). Element Interactivity and Intrinsic, Extraneous, and Germane Cognitive Load. *Educational Psychology Review*, 22(2), 123–138. <u>https://doi.org/10.1007/s10648-010-9128-5</u>

Sweller, J., Ayres, P., & Kalyuga, S. (2011). *Cognitive load theory*. Springer. https://doi.org/10.1007/978-1-4419-8126-4

- Sweller, J., Van Merriënboer, J. J. G., and Paas, F. (1998). Cognitive architecture and instructional design. *Educational Psychology Review*, 10, 251-295. https://doi.org/10.1023/A:1022193728205
- *Szymanski, M. H., Aoki, P. M., Grinter, R. E., Hurst, A., Thornton, J. D., & Woodruff, A. (2007). Sotto Voce: Facilitating Social Learning in a Historic House. *Computer Supported Cooperative Work* (CSCW), 17(1), 5–34. <u>https://doi.org/10.1007/s10606-007-9067-y</u>
- Thomas, A. E. & Müller, F. H. (2011). Skalen zur motivationalen Regulation beim Lernen von Schülerinnen und Schülern. Skalen zur akademischen Selbstregulation von Schüler/innen SRQ-A [G] (überarbeitete Fassung). (Scales of motivational regulation in student learning. Student academic self-regulation scales SRQ-A [G] (revised version). Wissenschaftliche Beiträge aus dem Institut für Unterrichts- und Schulentwicklung Nr. 5. Klagenfurt: Alpen-Adria-Universität.
- Tippelt, R., & Reich-Claassen, J. (2010). Lernorte: organisationale und lebensweltbezogenen Perspektiven. *REPORT - Zeitschrift für Weiterbildungsforschung, 104, 11-21.* <u>https://doi.org/10.3278/rep1002w011</u>
- Um, E., Plass, J. L., Hayward, E. O., & Homer, B. D. (2012). Emotional design in multimedia learning. *Journal of Educational Psychology*, 104(2), 485–498. <u>https://psycnet.apa.org/doi/10.1037/a0026609</u>
- Ungerer, L. M. (2016). Digital Curation as a Core competency in current learning and literacy: a higher education perspective. *The International Review of Research in Open and Distributed Learning*, *17*(5), 1-27. <u>https://doi.org/10.19173/irrodl.v17i5.2566</u>
- *Valle, N., Antonenko, P., Soltis, P. S., Soltis, D. E., Folk, R. A., Guralnick, R. P., Oliverio, J. C., Difato, T. T., Xu, Z., & Cheng, L. (2020). Informal multimedia biodiversity awareness event as a digital ecology for promoting culture of science. *Education and Information Technologies*, 25(4), 3275–3297. <u>https://doi.org/10.1007/s10639-020-10121-7</u>
- Van Gog, T., Kester, L., & Paas, F. (2011). Effects of worked examples, example-problem, and problemexample pairs on novices' learning. *Contemporary Educational Psychology*, *36*(3), 212-218. <u>https://doi.org/10.1016/j.cedpsych.2010.10.004</u>
- Van Gog, T., & Paas, G. W. C. (2006). Optimising worked example instruction: Different ways to increase germane cognitive load. *Learning and Instruction*, 16(2), 87-91. <u>https://doi.org/10.1016/j.learninstruc.2006.02.004</u>
- Van Harsel, M., Hoogerheide, V., Verkoeijen, P., & van Gog, T. (2019). Effects of different sequences of examples and problems on motivation and learning. *Contemporary Educational Psychology*, 58, 260–275. <u>https://doi.org/10.1016/j.cedpsych.2019.03.005</u>

- Wachsmuth, C. (2020). Computerbasiertes Lernen mit Aufmerksamkeitsdefizit: Unterstützung des selbstregulierten Lernens durch metakognitive Prompts. (Computer-based learning with attention deficit: supporting self-regulated learning through metacognitive prompts). Dissertation Technische Universität Chemnitz.
- Wahser, I. (2008). Training von naturwissenschaftlichen Arbeitsweisen zur Unterstützung experimenteller Kleingruppenarbeit im Fach Chemie. (Training of scientific working methods to support experimental small group work in chemistry). Dissertation. Universität Duisburg-Essen.
- Walker, J., Gibson, J., & Brown, D. (2007). Selecting fluvial geomorphological methods for river management including catchment scale restoration within the environment agency of England and Wales. *International Journal of River Basin Management*, 5(2), 131-141. https://doi.org/10.1080/15715124.2007.9635313
- *Wang, J., Yoon, S., Elinich, K., & Van Schooneveld, J. (2012). Investigating the Effects of Varying Labels as Scaffolds for Visitor Learning. In van Aalst, J., Thompson, K., Jacobson, M. J., & Reimann, P. (Eds.), *The Future of Learning: Proceedings of the 10th International Conference of the Learning Sciences (ICLS 2012)*, 1 (pp. 180-187). International Society of the Learning Sciences.
- Wellnitz, N. & Mayer, J. (2013). Erkenntnismethoden in der Biologie Entwicklung und Evaluation eines Kompetenzmodells. (Methods of knowledge in biology - development and evaluation of a competence model). Zeitschrift für Didaktik der Naturwissenschaften, 19, 315-345.
- Willems, A. S. & Lewalter, D. (2011). Welche Rolle spielt das motivationsrelevante Erleben von Schülern für ihr situationales Interesse im Mathematikunterricht? Befunde aus der SIGMA-Studie. In B. Schwarz, P. Nenniger, R. S. Jäger (Eds.), Erziehungswissenschaftliche Forschung – nachhaltige Bildung (pp. 288-294). Verlag Empirische Pädagogik.
- *Xambó, A., Hornecker, E., Marshall, P., Jordà, S., Dobbyn, C., & Laney, R.C. (2017). Exploring Social Interaction With a Tangible Music Interface. *Interacting with Computers, 29*, 248-270. <u>https://doi.org/10.1093/iwc/iww019</u>
- *Yoon, S.A., Anderson, E., Lin, J., & Elinich, K. (2017). How Augmented Reality Enables Conceptual Understanding of Challenging Science Content. *Educational Technology & Society, 20* (1), 156– 168.
- *Yoon, S.A., Elinich, K., Wang, J., & Schooneveld, J.G. (2012a). Augmented reality in the science museum: Lessons learned in scaffolding for conceptual and cognitive learning. *CELDA 2012*. 205-212.
- *Yoon, S. A., Elinich, K., Wang, J., Steinmeier, C., & Tucker, S. (2012b). Using augmented reality and knowledge-building scaffolds to improve learning in a science museum. *International Journal* of Computer-Supported Collaborative Learning, 7(4), 519–541. <u>https://doi.org/10.1007/s11412-012-9156-x</u>
- *Yoon, S. A., & Wang, J. (2013). Making the Invisible Visible in Science Museums Through Augmented Reality Devices. *TechTrends, 58*(1), 49–55. <u>https://doi.org/10.1007/s11528-013-0720-7</u>
- *Zimmerman, H. T., Gamrat, C., & Hooper, S. (2014). Connecting out-of-school learning to home: Digital postcards from summer camp. *TechTrends*, *58*(2), 87–92. <u>https://doi.org/10.1007/s11528-014-0739-4</u>
- *Zimmerman, H. T., Land, S. M., & Jung, Y. J. (2016). Using augmented reality to support children's situational interest and science learning during context-sensitive informal mobile learning. In A. Peña-Ayala (Ed.), *Mobile, ubiquitous, and pervasive learning: Advances in Intelligent Systems* and Computing, 406. (pp. 101-119). Springer International.

Appendix

Appendix A - Script Moodle learning environment

Introduction: Dear students, welcome to the water body structure mapping workshop. In the next 2.5 to 3 hours you will independently carry out a water body structure mapping. During this mapping you will support the steps of scientific observation. Please use this Moodle learning environment and the paper-based mapping sheet provided to you. For your individual learning outcome, it is important that you work on your own and do not exchange ideas with your fellow students! Please work on the following questions in order. Important! Please note that it is not possible to scroll back in the learning environment! Have fun and good luck with your mapping!

Next page:

Important! Please click on "Edit task" and enter the test person code you have created and noted on your sheets. Only go to the next page once you have entered and saved your test subject code!

Next page:

Water body structures are often required by municipalities and cities to assess a water body. Such a scenario now also forms the basis for your mapping. Please imagine the following: The Mühlengraben is a water body that flows through the municipality of Freising. The municipality of Freising is responsible for the maintenance of the Mühlengraben. The Mühlengraben is to be renaturalized in the course of flood protection, for which the structure of the water body is to be determined. The morphological characteristics that characterize a flowing water body (water bed and floodplain) or still water body (lake basin and banks with water environment) are described as water body structure. In the case of watercourses, these are in particular the form of the course (elongated, winding, branched), depth variation (pools, fords, banks, etc.), bed substrate and the characteristics of the bank areas. These are essential for the living conditions of animals and plants in and around the water bodies. In the Mühlengraben, in addition to improved flood protection, the main aim is to improve the quality of the habitat for all organisms that use the water body. In addition to the aquatic organisms in the canal, this also applies in particular to terrestrial organisms on the banks, which use the riparian strips that are now required by law. As there is only a very limited budget available for the renaturation measures, the first step is to draw up a water body development concept based on a water body structure mapping. In this water body development concept, all those sections with particularly poor structural quality are to be prioritized for renaturation. This water body structure mapping is to be carried out on behalf of the municipality of Freising by the planning office "Gewässergut" on the basis of the specifications of the Bavarian State Office for Water Management using standardized assessment forms. The stretch of water is divided into 100m long sections, each of which is examined and evaluated separately. These 100m sections will later serve as the basis for prioritization. The 100m sections with the worst watercourse structure are to be identified as the most urgent to restore. Following the water body structure mapping, the planning office should draw up a watercourse development concept of no more than ten pages, which reflects the results of the water body structure mapping, includes a prioritization of the stretches to be restored and explains your self-considered examples of measures in bullet points. You, as an employee of the planning office "Gewässergut", have now been instructed to carry out this water body structure mapping using the mapping form and the steps of scientific observation and then to draw up a well-founded water body development concept. Please go to the first mapping section and open the next page.

Next page: You are now in the first section of the water body to be mapped. When mapping, you will be guided by the steps of scientific observation, which will help you to determine the structure of the water body in a structured way. For the first two steps of scientific observation, the question and hypothesis, please watch the video first.

Task: After you have watched the video, please formulate a research question and one or more suitable hypotheses. To do this, click on "Edit task" again and write your answer in the field below. When you have finished, please click on "Save and display". Please note: Once you have saved your answer, you can no longer edit it! Important! Please make a note of the research question / hypothesis on your paper-based mapping sheet, as you will need it again later! Once you have completed the task, you can move on to the next page.

Next page: These two steps of scientific observation are about planning and conducting your water body structure mapping in order to answer your research question and your hypotheses. Please watch the video first.

Task: After you have watched the video, please carry out the mapping of the first water body section in the field using your paper-based mapping sheet. When you have finished mapping, please mark the task as "completed" and save your selection. The page will then reload and the "Next activity" button will appear, which will take you to the next page.

Next page: You have now reached the last two steps of the scientific observation, the evaluation of your observation results and the conclusion regarding the assessment of the mapping section (answering the research question and hypotheses). Please watch the video.

Task: After you have watched the video, please evaluate your data on the paper-based mapping sheet (answering the research question and hypotheses). Then click on "Edit task" and write your answer in the field below. When formulating your answer, remember that you are an employee of the planning office "Gewässergut" and should also make recommendations for action. Please remember again: Once you have saved your answer, you can no longer edit it! When you have finished, please click on "Save and display". You can then go to the next page.

Next page: You have now reached the end of the first mapping section. Please collect all your mapping sheets on your clipboard and start mapping the next section with a new sheet.

Next page: You are now in the second mapping section. Please watch the video again for tips on the research question. Task: After you have watched the video, please formulate a new research question and one or more suitable hypotheses. To do this, click on "Edit task" again and write your answer in the field below. When you have finished, please click on "Save and display". Please note: Once you have saved your answer, you can no longer edit it! Important! Please make a note of the research question / hypothesis on your paper-based mapping sheet, as you will need it again later! Once you have completed the task, you can move on to the next page.

These steps are now repeated until the fifth section.

End: You have now completed the water body structure mapping for all five sections and have reached the end. Please complete the questionnaire using the following link. When you have completed the questionnaire, please close the questionnaire window and mark the task as "completed" and save your selection. The page will then reload and the "Next activity" button will appear, which will take you to the next page.

Last page: You have now reached the end of today's workshop. Please hand in the iPad and clipboard to the person responsible and sign the list. Please check again that your subject code is on all five mapping sheets. We would like to thank you very much for your participation in the workshop "Mapping water body structure with the help of scientific observation" and wish you a wonderful day!

Appendix B - Script Video

Script Video Step 1 and 2 of the scientific observation

Theory: Every scientific observation begins with the development of a research question and a suitable hypothesis or several hypotheses. In our case, this involves the topic of water body structure. Because every scientific observation is based on a specific target or question as well as theoretically based expectations that can be formulated as hypotheses. But what makes a good research question? A good research question contains certain criteria that you should always consider when formulating your own research question. In general, your research question should be formulated in one sentence if possible and asked as an open question. This means that it should not be answered with "yes" or "no", but should be based on the W-questions such as "why, how, to what extent or in what way". An example would be "How do raindrops run down a window pane?" This could well be investigated on the basis of scientific observation. In general, when formulating your research question, it is important that your question is self-contradictory, can be answered in theory and is as objective as possible. In addition, the subject of your research should not be too extensive, but rather specific and clearly defined, as well as being observable within the framework of water body structure mapping. Once you have developed your research question, you must then deduce one or more hypotheses. But what are hypotheses? Hypotheses are well-founded assumptions that relate to your research question. Your research question is therefore tested by investigating the hypotheses. The same applies here as for the research question. Hypotheses must also be free of contradictions, answerable in theory and as objective as possible. They should be formulated as statements and include the conditions to be investigated. It is also important to formulate a alternative hypothesis, the null hypothesis, to contradict the hypothesis. If you have to reject your hypothesis, you can temporarily accept the alternative hypothesis.

Example: This all sounds quite theoretical, so I will now give you an example of a research question and corresponding hypotheses. You will see a photo of a section that is to be mapped. We will now think about a research question and the hypotheses, because that is exactly what you are supposed to do. When we look at the picture, we immediately notice the buildings and the closeness to the path, indicators that are mentioned on the mapping sheet under the criterion of riparian strip function. Assuming I want to focus on this in particular during the mapping, then my research question could be something like this: To what extent does development and the closeness of the path to the water body in particular have an influence on the water body structure? I would formulate my hypothesis as follows, for example: It is assumed that the development and the closeness of the path to the body of water have a negative influence on the water body structure and my null hypothesis would then be as follows: It is assumed that the development and the closeness of the path to the body of water do not influence the water body structure. It would therefore assume that it has no influence.

Tip: At the end, I have a tip for you on formulating your research question and hypotheses. Think about the different aspects that can be observed in the context of water body structure mapping. Also think about the mapping sheet that you have already looked at in preparation. Try to focus your research question on this. And now it's your turn, which research question do you want to evaluate? And which hypotheses do you assume? Good luck and see you in the next step.

Script Video Step 3 and 4 of the scientific observation

Theory: You have now thought about a research question and one or more appropriate hypotheses. The next two steps, planning and conducting the scientific observation, are about planning your own observation in detail and collecting observation data in order to test your hypothesis and answer your research question. The basis and tool for scientific observation is an observation instrument that contains precise instructions for defining the observation units and characteristics as well as for coding and recording the observation data. On the one hand, the observation instrument includes a defined

category system that is used to record the observation data. In addition to the precise description of the categories, a good category system also includes a precise differentiation of their characteristics, the so-called indicators. It must then be possible to assign each observation unit precisely to a category or its corresponding indicator. Therefore, the developed categories and their indicators must be disjunctive (i.e. have a clear distinction and no common element), exhaustive (assignment to a category must be possible in every case) and described with sufficient precision (assignment is based on precisely defined criteria, including examples). The exact definition of the categories and their indicators, as well as exemplary cases, must be described as precisely as possible in a so-called coding guideline. With the help of the guideline, you now know what is meant by the categories and their indicators and can assess to which category and to which indicator you can assign your observation data. With our water body structure mapping, you do not have to develop the observation tool yourself. You have already received the mapping and assessment instructions, which serve as a coding guide, as well as the assessment sheet for water body mapping with the category system including indicators.

Example: In the assessment sheet, for example, river bank stabilization is an upper category, with the subcategory quantity. The four indicators in this sub-category quantity are "no bank stabilization", "occasional", "moderate" and "predominant". You would now assign the result of your observation to one of the four indicators. Let's take a closer look at one category. On the left-hand image you can see the fourth section of a water body. On the right-hand side, I have shown you the criterion of transverse structures from the water body category. If I now look at the water body in the picture, I would say that no weir, fish ladder or similar can be seen in the picture. I would therefore mark the indicator not present.

Tip: Finally, I have a tip for you on planning and conducting your scientific observation. It is best to go through each evaluation category carefully one after the other, so you are sure not to forget anything! And now it's your turn to start mapping the first section!

Script Video Step 5 and 6 of the scientific observation

Theory: We now come to the last two steps, the data evaluation and the conclusion. You have now mapped the water body section with the help of the assessment sheet. Now it is time to evaluate and interpret the results. The evaluation and interpretation of the data is important, as the data obtained cannot stand for itself. The range of evaluations of observation instruments extends from simple recording of the frequency of appearance to more complex analyses; sometimes statistical evaluations and hypothesis testing are also included. Once the data has been evaluated, the aim is to describe the overall assessment as objectively as possible and without interpretation. Once the results have been described, the final step is to interpret and evaluate the data with regard to the hypotheses and the research question. In this step, it is also assessed whether the hypotheses can be regarded as preliminarily accepted or whether they must be rejected and the alternative hypothesis, the null hypothesis, is accepted. After you have accepted or rejected your hypothesis, you can then answer your research question. Once this has been answered, the resulting consequences and / or recommendations for action can be deduced and given. In our observation protocol, the individual indicators of water body dynamics in the respective categories are first added together and then summarized to form a structural class. The same procedure is then used to calculate the structural class of waterbody dynamics. Finally, the total structural class is determined on the basis of the two structural classes of water bed dynamics and floodplain dynamics. This figure then shows the overall assessment of the water body section to be mapped. You can see this in the image shown. This raises the question of what the number of the overall structural class actually says about the mapped water body section. The number of the structural class should be interpreted with regard to the hypothesis, for example, what does the number of the total structural class say? Is it a good rating (still in the blue or green range) or is the number already in the red range?

Example: Now let's take a look together at a fictional overall assessment. Let's assume that the riverbed structure is rated 6 and the floodplain structure is also rated 6. This results in a rating of 6 for the overall structure class, which shows that the overall rating of the water body structure is in the second worst range of 6 out of 7. We can therefore conclude that there is a need for action in both the river bed structure and the floodplain structure. To do this, we now need to take another look at the individual categories that led to this assessment. For the hypothesis we put forward at the beginning: It is assumed that the development and the closeness of the path to the water body have a negative influence on the water body structure, this means that we can provisionally regard it as accepted. We can now answer our research question: To what extent do the development and the closeness of the path to the water body in particular have an influence on the water body structure? in such a way that the development and the closeness of the path have a negative effect on the closeness of the path water body structure and the closeness of the path as a negative effect on the overall structure.

Tip: Finally, I have a tip for you regarding the evaluation and conclusion. Think again about what the individual points were that led to the evaluation. Were there any points on the mapping sheet that were weighted more heavily than others? Also use your knowledge from the introductory video.